

THE ROLE OF TEACHER LEADERS IN DEVELOPING AND LEVERAGING PROFESSIONAL LEARNING COMMUNITIES DURING AND POST THE COVID-19 PANDEMIC

BY

GLORIA KHULULIWE NGOAKO

A dissertation submitted in fulfilment of the requirements for the degree of Master of Education in the discipline of Educational Leadership Management and Policy

College of Humanities, School of Education

Edgewood Campus

February 2023

SUPERVISOR

Dr B.N.C.K. MKHIZE

DECLARATION

I, Gloria Khululiwe Ngoako declare that:

i. The research reported in this dissertation, except where otherwise indicated is my original

work.

ii. This dissertation has not been submitted for any degree or examination at any other

university.

iii. This dissertation does not contain other persons' data, pictures or other information,

unless specifically acknowledged as being sourced from other persons.

iv. This dissertation does not contain other persons' writing, unless specifically

acknowledged as being sourced from other researchers. Where other written sources have

been quoted then:

a) Their words have been re-written but the general information attributed to them has been

referenced;

b) Where their exact words have been used, their writing has been placed inside quotation

marks and referenced.

v. This dissertation does not contain text, graphics or tables copied and pasted from the

internet, unless specifically acknowledged, and the source being detailed in the dissertation

and in the references section.

	_
ign:	

SUPERVISOR'S AUTHORISATION

This dissertation is submitted with my approval.



Supervisor: Dr BNCK Mkhize

Date: 9 February 2023

DEDICATION

- I first of all, dedicate this work to the Almighty God, who protected, guided and provided me with a lifetime opportunity, spiritual strength, wisdom, courage and inspiration to complete this study. God gave me the power to triumph all challenges I faced throughout this research journey.
- I also dedicate this dissertation to my African queen, my mother Mrs Duduzile Nzuza (Gog' uNzuza), for being my prayer partner, 'mashonisa wam' and my pillar of strength.

 Thank you for teaching me the importance of education yet you are uneducated.
- To my older brother, Sipho Godfrey Nzuza, for being the father figure to me; he was, and still is on my side. The love he you has for me, humbles me.
- To my one and only son, Ndumiso Lwazi Mkhize and my only grandson, (for now), Lindokuhle (Okuhle) Kwande Mkhize, who are my reason for living. At times, I left them and stayed at UKZN till late but, still, they were patient with me.

ACKNOWLEDGEMENTS

I would like to express my sincerest gratitude to:

- My supervisor, Dr BNCK Mkhize, many thanks for his your patience, sacrifices, unwavering professional support and motivation hey has instilled in me to complete this work. I would like to thank him for nurturing my academic growth.
- All participants, teacher leaders, who agreed to partake in my study and for sharing their experience with me which assisted in the completion of this study.
- My encouragement and support system and the wind beneath my wings, my older sister Langelihle Khanyile, friends Ntokozo Mkize and Musakawukhethi Mshengu.
- My dearest family, my siblings, baby brother Sibusiso and baby sister Lungile, for their never-ending support. Guys the words you spoke and believe in me when kept me going.
- I also wish to express my appreciation to the KwaZulu-Natal Department of Education for allowing me to conduct this study despite difficult conditions imposed by COVID-19 pandemic.
- Last but not least, my church leaders Bishop Bhekimpi Mchunu and his wife Pastor Busisiwe Mchunu, for being the leaders who believe in education. You are educated yet you are studying further. You encourage everyone, young and old to study.

Thank you.

ABSTRACT

This research is aimed at exploring the role of teacher leaders during and post COVID-19 pandemic in Pinetown District in KwaZulu-Natal Province. The study focuses on teacher leaders' experiences in semi-rural schools during COVID-19 pandemic. This study explores the different experiences of the four teacher leaders from two schools. The objectives of this study are; to explore teacher leaders' role in developing and leveraging professional learning communities, to explore the challenges encountered by teacher leaders in developing and leveraging PLCs and explore appropriate strategies teacher leaders use to mitigate the challenges they encountered when developing and leveraging PLC in their schools during and post COVI-19 pandemic. This is a qualitative study located within the interpretive paradigm. The participants were purposively selected. The study used semi-structured interviews as a method of generating data. Due to COVID-19 regulations, interviews were conducted telephonically to observe COVID-19 protocols that forbid us from meeting face to face. The study employed thematic data analysis to analyse the generated data from the participants. The findings in this study revealed that the participating teacher leaders have a clear understanding of their role as teacher leaders. They however, endured varying experiences of their roles due to COVID-19 pandemic and the different contexts they work in. Some were still able to enact some of their instructional leadership experiences without any hindrances, while other participants experienced major COVID-19 related disturbances when enacting their instructional practices. The strategies used by these teacher leaders to mitigate the impact of COVID-19 pandemic included the following; drawing on teacher collaboration, professional development and workshops with different experts for skills development and traditional teachers to adapt to classes, resources and accomplishments of different version.

It is therefore recommended that teacher leaders should strengthen their efforts in ensuring that teachers work together. The study recommended the following; teacher leaders should attend workshops to equip themselves with technological skills and the latest technological teaching and learning models such as Microsoft Teams and Zoom Meetings; teacher leaders induced collaboration among their educators in order to ensure there was no interruption in the teaching, teacher-learner must be considered by employing the unemployed educators to minimise contact during teaching and learning and learning in their schools.

LIST OF ABBREVIATIONS

PLC(s) Professional Learning Community(s)

UKZN University of KwaZulu-Natal

TL Teacher Leaders

SMT School Management Team

HoD(s) Head of Departments

SPTD Senior Primary Teachers Diploma

MLMMS Maths Literacy Mathematics Mathematical Sciences

Bed Hons Honours in Bachelor of Education

EMS Economics Management Sciences,

ACE Advanced Certificate in Education

NCTM National Council of Teachers of Mathematics

HSSREC The Human and Social sciences Research Ethics Committee

ATP Annual teaching plan

PPE Personal Protective Equipment

ICT Information Communication Technology

TABLE OF CONTENTS

CONTENT	PAGE
Declaration	i
Supervisor's statement	ii
Acknowledgements	iiii
Dedication	iv
Abstract	v
List of abbreviations	vi
Table of contents	vii
Appendices	xi
CHAPTER ONE	
INTRODUCTION AND BACKGROUND	
1.1 Introduction	1
1.2 Background of the problem	1
1.3 The research problem	3
1.4 Study objectives	4
1.5 Research questions	4
1.6 Rationale and motivation for the study	4
1.7 Significance of the study	6
1.8 Clarification of key concepts	7
1.8.1 Leadership	7
1.8.2 Teacher Leadership	7
1.8.3 Professional Learning Community	8
1.8.4 COVID-19 Pandemic	9
1.9 Study limitations	9
1.10 The overall structure of the study	10

11

CHAPTER TWO

2.1 Introduction

2.2 Conceptualising teacher leadership	11
2.3 Conceptualising PLCs	12
2.4 The role of teacher leaders in developing PLCs	12
2.4.1. The leader and the facilitator	12
2.4.2. Promoting coherency and connecting	13
2.4.3. The skills developer	13
2.5 Leveraging on PLCS	14
2.5.1 Promoting collaborative learning	14
2.5.2 Creating an online teaching and learning platforms for collaborative learning	15
2.5.3 Building school capacity through PLCs	16
2.5.4 Professional evaluation reflective and feedback	16
2.6 Barriers to developing and leveraging PLCs	17
2.6.1 Lack of adequate support from departmental structures Lack of resources for the successful implementation of online education	17
2.6.2 Resistance to change	19
2.7 Strategies to mitigate barriers encountered by teacher leaders when developing and leveraging PLCs	19
2.7.1 Generosity, appreciation and compassion	20
2.7.2 Setting sensible expectations for colleagues and having tolerance for other people and self	21
2.7.3 Creating positive philosophy to produce professional abilities, capabilities and skills to enhance and broaden other leaders' expertise/Cultivating and growing other leaders	22

2.7.4 Teacher leadership as the web of teacher collaboration	23
2.8 Theoretical framework	
2.9 Chapter summary	
CHAPTER THREE	
RESEARCH DESIGN AND METHODOLOGY	
3.1 Introduction	27
3.2 Research paradigm	27
3.3 Research approach	28
3.4 Research design	29
3.5 Selection of participants	30
3.6 Data generation method	30
3.7 Data analysis	31
3.8 Ethical Issues	32
3.9 Issues of trustworthiness	32
3.9.1 Credibility	33
3.9.2 Transferability	33
3.9.3 Dependability	33
3.9.4 Confirmability	34
3.10 Chapter Summary	35
CHAPTER FOUR	
DATA ANALYSIS AND DISCUSION	
4.1 Introduction	36
4.2 Profile of the participants	37
4.3 Data presentation and discussion	38

4.3.1 Teacher leaders' understanding of professional learning communities	38
4.3.2 Teacher leaders' role in developing PLC at the school	40
4.3.3 The impact of COVID-19 on the functioning of PLCs in schools	41
4.3.4 Promoting coherency and connecting opportunities for teacher learning in PLCs during this pandemic	46
4.3.5 Challenges encountered in developing PLC during this time?	48
4.3.6 How to 'deprivatise practice' and prevent teacher isolation with the COVID-19 protocol in place	50
4.3.7 The means or strategies tried to leverage PLCs during the pandemic	52
4.3.8 Maintaining shared routines and the way forward for PLCs post the pandemic,	56
4.4 Chapter Summary	60
CHAPTER FIVE	
SUMMARY, CONCLUSIONS AND RECOMMENDATIONS	
5.1 Introduction	
5.2 Study summary	61
5.3 Presentation of Findings	62
5.3.1 Participants" understanding of professional learning communities (PLC) and their role in developing PLC at the school	62
5.3.2 The impact of COVID-19 on the functioning of PLCs and promoting coherency and connecting opportunities during this pandemic	62
5.3.3. Challenges encountered in developing PLC and 'deprivatise practice' and prevent teacher isolation with the COVID-19 protocol in place during this time?	63
5.3.4 The strategies to leverage and maintaining shared routines and the way forward for PLCs post the pandemic.	64
5.4 Recommendations	65
5.4.1 Consolidating technology in teaching and learning	66
5.4.2 Supporting reasonable teacher-learner ratio	66

5.4.3 Traditional teachers to adapt to classes, resources and accomplishments of different version	
5.5 Chapter Summary	68
APPENDICES	PAGE
APPENDIX A: Permission letter to KwaZulu Natal DoE	93
APPENDIX B: Permission letter to school principals	95
APPENDIX C: Permission letter to participants	97
APPENDIX D: Declarations	99
APPENDIX E: Interview schedules	100
APPENDIX F: Turnitin certificate	101
APPENDIX G: Ethical clearance certificate	102

CHAPTER ONE

INTRODUCTION AND BACKGROUND TO THE STUDY

1.1 Introduction

In the first quarter of 2020, South Africa experienced the emergence of COVID-19 pandemic. The pandemic put constrains on schools and the education sector in the country responded by closing schools unprecedentedly. The schools' day-to-day operation was disrupted because of the spread of COVID-19 globally (Cheng, 2020; Mathew & Mahoney, 2020). As the crisis continued, school leaders had to act and act fast to redeem the situation and save the schooling systems. School leaders had to explore short-term and long-term strategies to provide for distance learning and for support to their teams (Bennett, & Bromen, 2019). They had to work hard to develop others' sense of agency, while on the other hand, they had to develop themselves as well (Nguyen, & Hunter, (2018). The teacher agency was also visible in addressing these issues. However, it cannot be said with any sense of confidence that teacher agency was visible in all schools. This study focused on exploring how teacher leaders developed and leveraged professional learning communities during and post the outbreak of COVID-19 pandemic to enhance professional development in their schools and keep the process of teaching and learning running.

This is the first chapter of the study, and it serves as an orientation to understanding the research problem. The chapter begins by providing a background to the problem, followed by a formal declaration of the statement of the problem. Thereafter, the study objectives follow. This is followed by the research questions and the rationale and motivation for the study. Other usual elements of the first chapter such as the significance of the study, the demarcation and limitations of the study follow. Towards the end of the chapter, an outline of the whole study and chapter summary concludes the chapter.

1.2 Background to the problem

When the COVID-19 pandemic came into existence, staff meetings, corridor laughs and talks disappeared amongst teachers. Because of the restrictions brought by the pandemic, teacher leadership roles were embraced by moving beyond traditional implementation techniques (Naidu & Laxman, 2019; Vermeulen et al., 2016; Yuen & Ma, 2008). Some education institutions worldwide embarked on making use of online teaching and learning (Coman, Tîru,

Meses, an-Schmitz, Stanciu, & Bularca, 2020). Many teachers had no past experience in using online teaching (Bozkurt & Sharma, 2020). This posed a challenge to several teachers who were now isolated due to social distancing regulations and a number of other restrictions imposed by the government. School leaders had to think outside the box to keep teachers motivated and to embrace online teaching despite challenges which came with it. For example, many teachers had no past experience in the use of online teaching. As inexperienced as they were, they needed school leaders to provide ongoing professional development, or the space where they could meet and discuss formerly or informally, how to deal with the situation. Teacher leaders are a necessary structure to support educational improvement efforts (Kajee, 2011).

Teacher leaders have the responsibility to initiate things and should be willing to work collaboratively with other teachers (Grant, 2012; Riveros, Newton, & da Costa, 2013); advance professional skills, competence and knowledge of other teachers (Abbott, 2014). Teacher leaders should take chances in initiating new things in order to display their teaching skills and should be creative thinkers in solving problems (Danielson, 2006). The South African Schools Act, No. 84 of 1996, makes provision for teacher leaders to take leadership roles in respect of the subject, learning area or phase if required, collaborate with teachers in organising and conducting extra and co-curricular activities (Republic of South Africa, 1996). Thus, in the unprecedented situations like the one during the pandemic, teacher leaders had mounting leadership responsibilities such as decision- making, mentoring, coordinating and inspiring colleagues to sustain quality in teaching and learning (source). It is therefore, the expectation that even in the crisis brought about by the COVID-19 pandemic, teacher leaders would be at the forefront to restore normality in teaching and learning, take affirmative measures to set instructional culture of the schools, follow COVID-19 regulations and protocols to create conducive environment for teaching and learning, and boost the morale of teachers and learners knowing that schools are deemed as hotspots for the widespread of infectious diseases, which was the main reason for the closure of schools worldwide (Abdulamir & Hafidh, 2020).

1.3 The research problem

Teacher leadership is objectively restricted in South African schools (Chatturgoon, 2009;, Grant, 2006; Rajagopaul, 2007; Singh, 2007). The South African education policy does not simplify what teacher leadership involves, nor does it offer strategies on how teacher leadership

should be presented in schools (de Villiers, 2011). It is contended by Fairman and McKenzie (2012) that environments conducive to teacher leadership and development seem to be absent in some schools. Other studies (Grant, 2006; Grant & Singh, 2009; Grant, 2012; Monametsi, 2015; Naicker & Somdut, 2014) suggest that not much support is provided in developing and advancing teacher leadership in South Africa. Similarly, (Grant, 2006, 2008, 2017; Khumalo, 2008) demonstrate that teacher leadership practice is happening at a very slow pace and is not keeping with the new demands policy demands. Evidence emerging from Grant's review of South African studies of teacher leadership suggests that teachers are severely restricted in terms of the degree to which they could become involved in organisational leadership (Grant, 2017). With these restrictions, the question arises as to how teacher leaders developed and leveraged professional learning communities during and post the outbreak of COVID-19 pandemic to enhance professional development in their schools.

Since the outbreak of COVID-19 pandemic in South Africa in early March 2020, teacher leaders have been challenged by the precise and demanding regulations and protocols promulgated to curb the spread of this virus (Abdulamir & Hafidh, 2020). As a result of the surge of this virus, teacher leaders are confronted with the conundrum of adapting their known and expected leadership practices to lead in these unprecedented changes with the aim of achieving quality teaching. There seems to be limited research on how these accelerated COVID-19 related changes impacted on how the teacher leaders developed and leveraged professional learning communities during and post the outbreak of COVID-19 pandemic to ensure that the quality of teaching and learning is maintained in schools. Little is known about how teacher leaders, for example, take affirmative measures to set instructional culture of the schools, follow COVID-19 regulations and protocols to create conducive environment for teaching and learning and boost the morale of teachers. This pandemic has highlighted the need for school stakeholders to collaboratively work closely together for the betterment of the learners (Myende & Nhlumayo, 2020). This study sought to explore how teacher leaders developed and leveraged professional learning communities during and post the outbreak of COVID-19 pandemic. It is worth documenting and learning from teacher leaders' experiences of how they confronted the conundrum of adapting their known and expected leadership practices to lead in these unprecedented changes for present and future reference.

1.4 Study objectives

The aim of this study was to explore how teacher leaders developed and leveraged professional learning communities during and posts the outbreak of COVID-19 pandemic and seeks to achieve the following objectives:

- To explore teacher leaders' role in developing and leveraging professional learning communities in their schools during and post the COVID-19 pandemic.
- To explore the challenges encountered by teacher leaders in developing and leveraging PLCs in their schools during and post the COVID-19 pandemic.
- To explore appropriate strategies that teacher leaders use to mitigate the challenges they encounter when developing and leveraging PLC in their schools during and post COVI-19 pandemic.

1.5 Research questions

The study was guided by the following research questions:

- What are the teacher leaders' experiences of their role in developing and leveraging professional learning communities in their schools during and post the COVID-19 pandemic?
- What challenges did teacher leaders encounter in developing and leveraging PLCs in their schools during and post the COVID-19 pandemic?
- What appropriate strategies do teacher leaders use to mitigate the challenges they encountered when developing and leveraging PLC in their schools during and post COVID-19 pandemic?

1.6 Rationale and motivation for the study

The rationale for undertaking this study has binary dimensions that influenced it; the personal and academic. In my twenty-eight years teaching experience and experiencing lots of curriculum and other changes in the education sector, COVID-19 was the most drastic change I have experienced ever. When COVID-19 first struck, I was in the dark. I did not know what to expect and what to do. In the midst of long list of the things to do and not to do provided by the government and the Department of basic Education, my main concern was the future and

the success of teaching and learning. Encamped by fear and confusion, I waited for the education sector to pave the way forward. I needed to be empowered with the appropriate technological skills, knowledge and expertise due to the introduction of online teaching and learning at the time when staff meetings, corridor laughs and talks disappeared amongst teachers. This increased my interest to explore how teacher leaders would promote professional learning communities to provide the setting and necessary support for classroom teachers to participate collectively in development activities.

The schools' genuine objective was to create credible teaching and learning practices in line with COVID-19 guidelines for schools to enable teacher preparation programmes. It was not easy to adopt the new preparations programmes. When we return to schools, the school life had radically changed, both socially and professionally. School life was not the same as it was when we left before COVID-19 struck. Social learning opportunities which exist in the workplace are essential to brainstorm ideas, to discuss teaching practices, and to problem-solve situations (Gerken, Beausaer, & Segers, 2016). Teacher leaders in my school seemed to have no platform to help teachers with online teaching because of the social distancing policy which was highly exaggerated by some staff members. Technology implementation has been shifting not only the way communities work but also how education forms our current generations. Digital contexts have become indispensable tools to work, communicate and engage with individuals from professional to personal environments (Schmidt, Tschida, Christina, & Hodge, 2016). While the challenges facing the country's education system are diverse, solutions need to be specific, clear, sustainable, and relevant (Beaunoyer; Dupéré; Guitton, 2020). In tackling some of these barriers, one wonders how the teacher leaders could play a meaningful role to teacher development.

The March 2020 COVID-19 social distancing practices resulted in the abrupt, required transition to online learning and the dependence on instructional technology. However, having access to technology is not enough when educators find themselves with an opposing perception towards technology due to a lack of knowledge or digital skills (Dinc, 2019; Vongkulluksn, 2018). Educational technology in the classroom might leave a negative impact on teacher preparation if there is an unforeseen transition from regular methodologies to the implementation of digital tools due to the lack of support and knowledge (Vongkulluksn, Xie, & Bowman, 2018), which occurred during the pandemic transition.

Over the last two decades, research on teacher leadership in South Africa has gained momentum (Grant, 2019; Makoelle & Makhalemele, 2020; Naicker et al., 2016). It has gained momentum to the point that teacher leadership models have begun to develop. For example, Grant (2008) developed a teacher leadership model that encompasses four zones where teachers should be active participants in order to be called effective teacher leaders. Other researchers have interrogated the enablers and constraints to teacher leadership practices in South African schools (Blose & Khuzwayo, 2020; Naicker et al., 2016; Webber, 2021). The extant research literature reveals several mechanisms of successful teacher development, including the approach of in-service co-learning through a system of professional learning communities (Feldman, 2020). However, this does not occur naturally, and requires school leaders to develop and support it. The literature I have engaged with demonstrated complications in the teacher leaders' understandings of their leadership roles owing to complex nature of various contexts. One can imagine that the complexity and uncertainty brought by COVID-19 pandemic does not make it particularly easy for teacher leaders to perform their duties to the best of their abilities to ensure that effective teaching and learning still takes place in schools. That is where my interest arose from, and I was keen to explore the leadership experiences of teacher leaders during COVID-19 pandemic. Therefore, this study sought to explore how teacher leaders developed and leveraged on professional learning communities during this unprecedented time. There is paucity of research exploring the concept of professional learning communities and the role that can be played by teacher leaders in developing and leveraging on PLCs during and beyond COVID-19 crisis.

1.7 Significance of the study

The intended outcome of this research is to explore how teacher leaders develop and leveraged professional learning communities in their schools during and post the COVID-19 pandemic. Therefore, it is crucial that we document and learn from their experiences. These lessons could be useful to other teacher leaders working in similar contexts where they are expected to effectively lead during a time of crisis such as COVID-19. Through this practice, current and aspiring leaders can focus their efforts on aspects of the job that are likely to be most effective and efficient in improving teaching and learning. Also, the findings generated from this study could act as a blueprint of understanding teacher leadership that might contribute to the improvements in the instructional leadership culture of the school during an unprecedented

time. Departmental officials at circuit and district levels responsible for policy formulation may benefit from this study since it can assist them in designing relevant policies and mechanisms for teacher leaders to improve their leadership skills when confronted by uncertainty.

1.8 Clarification/Definition of key concepts

There are four key concepts that are clarified in this section, and these are leadership; teacher leadership; professional learning community and then COVID-19 pandemic.

1.8.1 Leadership

The concept 'leadership' has been widely researched and is often seen as a contested concept because of the different views that researchers have taken in attempting to give leadership a definition (Bush, 2015; Bush & Glover, 2016). Leadership is about vision and about leaders working with people to shape motivations, actions, and goals to move the organisation in the right direction (Andriani, Kesumawati, & Kristiawan, 2018). As part of working with people to move the organisation forward, leaders need to influence and direct people to achieve the vision set out, particularly when change is being implemented (Bush, 2015; Bush & Glover, 2016). I use leadership to refer to anyone who influences other colleagues in achieving the school goals.

1.8.2 Teacher Leadership

There are formal teacher leaders who apply, are chosen and selected for positions and obtain training for their accountabilities; and casual teacher leaders who arise impulsively from the teacher positions (DeMore Palmer, 2011). Formal leaders as I understand them are those who form the School Management Team (SMT) in a school, namely, the school principal, the deputy principal and departmental heads. Conversely, informal teacher leaders are PL1 teachers who spontaneously play leadership roles within and outside the classrooms and in the community. The attention of this study is on PL1 educators as teacher leaders and how they enact teacher leadership in schools. Jackson, Burns, Bassett and Roberts (2010) indicate that teacher leaders should have leadership, vision, positive effect, work ethic, openness, team work

and risk taking and teacher related skills for them to be recognised as teacher leaders. Again, Jackson et al. (2010) recognised the requirement for teacher leadership to improve capacity to work with other teachers as part of teacher leadership enactment. Teacher leadership in this study refers to teachers or leaders working with colleagues to achieve a vision that enables the advancement of teacher leadership, which enhances student achievement. I also use leadership to mean teachers that use their skills, energy and creativity to enhance teacher leadership within their schools by influencing, directing and motivating their colleagues to participate in leadership tasks that champion teaching and learning. Teacher leaders take chances in initiating new things in order to display their teaching skills and they are also creative thinkers in solving problems (Danielson, 2006; Katzenmeyer & Moller, 2001). Teacher leaders are expected to function in the following areas: in the classroom engaging in teaching and learning, outside the classroom engaging with their colleagues and learners, as part of the whole school development and as a community member (Grant, 2012).

1.8.3 Professional Learning Community

The second concept that is briefly discussed is Professional Learning Community (PLC). This concept is a professional development framework that helps to create a sense of community as well as offering a space where teachers can collaborate and learn together (Banasik & Dean, 2016). Panich (2016) describes PLC as a continuous process where teachers leaders together with teachers work in the cycle of common questioning and learning to achieve better learning outcomes for themselves and for the learners (Bakar & Jamian, 2016). It is used as a platform for planning, strategising to tackle problems and building teaching confidence. Moreover, PLC has proven to have positively impacted teachers' professional development and school improvement (Harris & Jones, 2017)). During this time of the pandemic outbreak, PLC can be is used as the skills developer for teachers who are technologically challenged. DuFour (2006) explains that to accomplish a successful PLC, it is necessary to integrate three elements, namely,: setting up goals for student learning success by identifying their different needs, creating a strong sense of collaborative culture amongst teachers, and finally, teachers should work together to reach results based on continuous up-to-date teaching practices. Therefore, a positive school environment is created by effective PLC where teachers interact and communicate with one another, exposing barriers and aligning digital tools with their instructional goals (Baran, 2016).

1.8.4 COVID-19 Pandemic

According to Du Toit and Anadolu (2020), COVID-19 started in Wuhan City, Hubei Province of China (Original epicentre of COVID-19) and spreading around the world in less than 3 months. COVID-19 pandemic is considered as one of the biggest pandemics that have ever been found among humans. The medical features of COVID-19 are well documented, with most people displaying severe acute respiratory symptoms or mild symptoms or none at all but they ended up being dead because of COVID-19. This occurred mainly to the elderly and chronically ill patients. The spread of COVID-19, among several disruptions to normal life, necessitated more than 160 countries to effect temporary closure of schools. The World Bank (2020) estimates that the closure of schools left 1.6 billion children and youth out of school.

In South Africa, the government was forced to enact a national lockdown, which meant that there was total closure of all schools, including universities. This caused a halt to the learning process. There is concern among some in society that the widespread school closures would lead not only to the loss of learning, but also to the loss of human capital and diminished economic opportunities in the long run (The World Bank Rapid Response, 2020).

1.9 Delimitation/Demarcation of the study

This study was conducted in two of the schools in the Pinetown District. The schools where the study was conducted are located in a densely populated and is known to have a high rate of unemployment. Most of our learners' parents have primary school education or no education at all. The study comprised four teacher leaders. The participants were two teacher leaders, leaders per school, and were drawn from different phases in each school. Two teacher leaders were from primary school and the other two were from the secondary school in the same community. The focus of the study was on their role in developing PLC during and beyond COVID-19 pandemic.

1.10 The overall structure of the study

This study is made up of five chapters and in this section I sign-post the route I have taken in compiling this study report. Chapter one is about the background and the orientation of the role of teacher leaders in developing and leveraging professional learning communities in their

schools during and posts the COVID-19 pandemic. The purpose and rationale for the study is outlined. The significance of the study is explained. Aims and objectives are identified with key questions that the study sought to answer. Chapter Two presents the literature review and theoretical framework of the study on teacher leaders and how they develop and explore barriers and strategies to mitigate barriers through professional learning communities. In this chapter, I unpack critical concepts with regard to the study, scholarly debates on teacher development and learning communities during and beyond COVID-19.

Chapter Three describes the research design of the study, an overview of the empirical study, the approach, research design and methodology, including sampling procedures, the selection of participants, the discussion of ethical issues, limitations and trustworthiness and how rigour was exercised. Chapter Four discusses, and analyses the data produced with regard to the research topic. As part of the analysis and the discussion, I injected both local and international literature as well as theoretical frameworks. Chapter Five presents the results of the analysis of features of the teacher leaders in professional learning communities and how it relates to the findings of the study. In presenting the results, I begin with the summary of study, draw conclusions of the study and make recommendations.

1.11 Chapter Summary

This chapter deals with the introduction and overview, it includes the purpose and the rationale of the study. Critical questions and objectives of the study are listed as well. Key concepts are defined. Lastly, delimitations of the study are highlighted as well as how the outline of the study was organised. Literature review including conceptual frameworks is used in the study. The next chapter presents a review of literature and the theoretical framework of the study.

CHAPTER TWO

LITERATURE REVIEW AND THEORETICAL FRAMEWORK

2.1 Introduction

The previous chapter provided an orientation and a direction to the study. A layout of all critical parts of the research process was talked about. This chapter discussed international and local literature on the role played by teacher leaders in the development of PLCs during and after COVID-19. Moreover, this chapter deliberated on how to create an online community of teaching learning, the barriers encountered as well as mitigation of the barriers encountered. I begin the discussion of the review of literature by conceptualising teacher leadership. This is followed by other key aspects of this chapter such as conceptualising Professional Learning Communities, the role of teacher leaders in developing Professional Learning Communities, leveraging Professional Learning Communities, barriers to developing and leveraging Professional Learning Communities, strategies to mitigate barriers encountered by teacher leaders when developing and leveraging Professional Learning Communities. Towards the end of the chapter, I discuss the theoretical framework underpinning the study.

2.2 Conceptualising teacher leadership

Leadership includes both assigned leaders, who hold formal leadership positions within an organisation as well as emergent leaders, those who do not hold formal leadership roles but are influential to group or organisational members and of whom group members are accepting of this influence (Northouse, 2019). Teacher leaders must frequently communicate with all stakeholders to solve different matters related to teaching and learning utilising distinctive correspondence channels. The main reason for teacher leaders is to ensure that teaching and learning and is smoothly taking place and is mostly productive. Teacher leaders and school management team are ordinarily considered liable for of their learners academic achievements (Busch, O'Brien & Spangler, 2005). They ensure smooth interaction amongst teachers and they work hand in hand with the school management team.

Schott, van Roekel, and Tummers (2020) refine the concept of teacher leadership in three-fold: (1) teacher antecedents which involves teachers' skills, attributes, knowledge, and personalities. (2) Schools' antecedents which involves the principal and his SMT and how they provide support and growth opportunities. At times it may include peers' role-modelling or organisational policies: (3) Supra-school level antecedents which are a pre-service training, professional learning and governmental certification.

2.3 Conceptualising Professional Learning Communities

Basically, within the school context, PLCs can be viewed as including professionals enabling teachers to continually acquire from one another through shared visioning and organising, and evaluating what to achieve and that there is less in improving learners achievement (Opfer & Pedder, 2011). The notion of the PLC seems to be the most outstanding professional development and adjustment routine in many schools. When there are functional PLCs, these leads to visible improvement in learners' learning (DeWitt, 2016).

2.4 The role of teacher leaders in developing Professional Learning Communities

The role of teacher leaders in developing professional learning communities is discussed under three themes, namely, the teacher leader as a leader and a facilitator; promoting cohesion and connecting, as well as the teacher leader as a skills developer.

2.4.1. The leader and the facilitator

According to Wenner and Campbell (2017: p 134-171), "teacher leaders have the capacity to lead the school via increasing teacher collaboration, spreading best practices, encouraging teacher professional learning, offering assistance with differentiation, and focusing on content-specific issues". This includes both formal and informal roles as teacher leaders inside and outside the classroom. Teacher leaders have dual roles that of being teachers and also being leaders. Supporting this assertion, Shen et al. (2020), describes teacher leaders as advocates of learning beyond their classrooms and can influence and inform improvement in their respective schools. In the context of this study transitioning from face to face teaching and learning to online/digital teaching and learning, their responsibilities will surely increase.

Developing PLCs in semi-rural and mostly under resourced schools would be "a process of influencing others rather than a collection of characteristics or tasks of so-called teacher leaders" (Scott, Roekel, & Tummers, 2020: p. 100352). They have their own subject that they teach whilst managing others. They are entrusted with extra responsibilities that go beyond collaboration implementation, leading teams of people, managing resources, solving problems,

being innovate and developing teachers' confidence and interdependency in their respective sectors (Shaked & Schechter, 2017). According to Chiniara and Bentein (2016), teacher leaders support their colleagues' need for independence by enabling them to take the initiative, learn from mistakes, and handle difficult situations in their own way. The sudden change that COVID-19 pandemic brought must be overcome by facilitating teachers' PLCs and professional developments' support (Reimers & Schleicher, 2020).

2.4.2 Promoting cohesion and connecting

The National Council of Teachers of Mathematics (NCTM) (2016) highlights that "coherence means that connections are made from one year to the next, from one idea to another, from one representation to another. For a school to be successful in needs effective leadership and management. Effective teacher leaders provide good learning opportunities for learners. High quality leadership makes a meaningful school improvement and learning outcomes. Webber and Nickel (2021; p 17-49) suggest "teacher leaders must necessarily seek a sustainable balance in their commitment to the multiple educational priorities that compete for their attention". For this study because of the impact of COVID-19 teacher leaders tried to avoid the top-down traditional ways of professional development. Traditional ways are often occasional and unclear at times to impact teachers. Hence, cohesion and connection is not happening because teachers seem to be followers not owners of developments.

A peer centred way of doing things is implemented easily by teachers. Creating more coherence and connected opportunities through peer centred way is not automatic but it can be developed in any way suitable for them. Teacher leaders are responsible for encouraging and motivating teachers to work together and work hard through information sharing. Teacher leaders connect with PLCs through service, influence, leadership, and informal power aspect (Cansoy & Parlar, 2018).

2.4.3 The teacher leader as a skills developer

In the 21st century, there is a growing realisation that headship is a specialist occupation that requires specific preparation. Effective preparation is one way of reducing the 'shock' and helping leaders to cope. In some instances teacher leaders and teachers have varying degrees

of online-based facilitation and learning experience. Chothia (2020) states that schools need to use the lockdown break to explore digital and online delivery methods for teaching and learning to support programmes at a later stage. In the meantime schools, teachers and learners in South Africa have inadequate or no access to computers, laptops and cell phones, which are necessary for online teaching and learning, especially in the rural and under resourced communities (Bhengu, 2021, Black, Spreen & Vally, 2020). As lockdown eased in some countries, the convenors of informal educational activities faced unexpected costs to ensure the safety of participants and facilitators. O'Toole's (2018) work found that teachers are a type of first responder in a crisis – but unlike other types of first responders, they are not trained to work safely or effectively in this capacity. In this case, O'Toole (2018) also noted that teachers shared in the trauma of the incident – having their own lives and families impacted. The idea of providing the right resources and motivation in a crisis is not unique to COVID-19 cases.

The work of teacher leaders during difficult times can be daunting and exhausting. Teacher leaders' exhaustion escalated where schools resources were not provided to help educate/skill teachers technologically and giving them emotional support to deal with high load of family priorities, apprehension around finances and job security. Fletcher and Nicholas (2016) maintain that it is important for leadership to be flexible in improvising and adjusting throughout the crisis and being prepared for times ahead. However, one cannot say that there will not be blunders and amendments along the way.

2.5 Leveraging Professional Learning Communities

The discussion about leveraging on professional learning communities is presented under four themes, namely, promoting a collaborative learning culture; creating an online teaching and learning platforms for collaborative learning; building school capacity through PLCs and professional evaluation reflective and feedback. These are discussed next.

2.5.1 Promoting a collaborative learning culture

"Studies show that promoting curriculum change via teachers can be best achieved through utilizing teacher development and through stimulating collaborative curriculum design" (Garet, Porter, Desimone, Birman, & Yoon, 2001; Penuel, Fishman, Yamaguchi, & Gallagher, 2007). Teacher collaboration is an important piece of developing both a shared definition and coherence. The benefits of teachers collaboratively generating new knowledge about curriculum and teaching in schools or teams are increasingly recognised (Bhengu & Blose, 2022; Cober, Tan, Slotta, So & Könings, 2015; Cochran-Smith & Lytle, 1993, 2009; Hubers, Poortman, Schildkamp, Pieters & Handelzalts, 2016; Pareja, Roblin, Ormel, McKenney, Voogt & Pieters, 2014).

In context of this study, I regard collaboration as a way that will make transition easy-going. As much as collaboration is a product activity, it is not a forceful activity, it is voluntary (Wilson, 2016). All in all, school leaders and teacher leaders must help all teachers in the school to be part, to be attached and be committed to collaboration. A few different ways to do this including articulating a reasonable, explicit and convincing vision; coordinating with assignments and roles to staff individuals who are actually put resources into them; extending influential positions; and making coordination simple. Occasionally, teachers make excuses for not collaborating. The truth is that they have tight schedules at hand and they may find it easy to work in isolation than working in collaboration which is as productive as collaborating (Wilson, 2016).

2.5.2 Creating online teaching and learning platforms for collaborative learning

Collaborative learning is viewed as a fundamental supporter of active learning and colleagues share information and exchange knowledge; it also promotes learning which improves learners' effectiveness as well as advances learning performance (Chen, Law & Chen, 2017). The implementation of digital proficiency and the change of required technological skills in teacher leadership roles (Van Wart et al., 2017). The Glossary of Education Reform, (2014) speaks of vertical coherence and horizontal coherence. Vertical coherence it is when teachers teach the same programme in a grade, horizontal is when same grade teachers collaborate to align their learning activities, instruction, and assessments. Before the arrival of COVID-19 teacher leaders depended on curriculum resources and standards but during COVID-19 they were forced to rely on online resources to connect vertically and horizontally (Djalante, Lassa,

Setiamarga & Rafliana, 2020). These ways make teachers leaders and PLCs to collaborate to develop a high-quality coherent curriculum (Chalk, 2020).

Zhong (2017) discusses digital teacher leadership roles as modelling and of creating a shared vision. Teacher leaders and PLC members can use technology to teach, assess learners and as the tool for teacher development. Concurrently Leithwood and Jantzi's (2000) highlighted that for a school to be successful during the transformation period purposes and goals must be absorbed by teachers and formal leaders of the school. Nowadays online-based PLCs importance is more of a professional work to teach, assess and record keeping, and these considerations are done online. When learners and teachers obtain the appropriate assistance for digital online learning, the results are mostly outstanding.

2.5.3 Building school capacity through Professional Learning Communities

'We are all in the same boat' is a phrase that is used when people work hand in hand. Commitment to professional roles, colleagues' growth, coaching, and mentoring are a collective responsibility. In order for the PLCs to be effective in schools, school leadership must be part of the shared responsibilities for school and staff development. Using digital information must become a top priority in schools. School leadership through their commitment to schools' development must support teacher technology skills training together with teacher leaders. Around the world, teacher leaders have thrived to obtain solutions and support learners' learning through different appliances (Netolicky, 2020; Rasmitadila et al., 2020). At times COVID-19 reports can be demoralising to teacher leaders and their PLCs but working synchronously with school leadership gives relief. For a school to smoothly usher in online teaching and learning, school leadership must also regard themselves as facilitators of PLCs between members by maintaining effective communication among teachers. This helps the school, teacher leaders and PLCs to discover 'hidden potentials' and expertise of novice teachers. They can be encouraged and develop their abilities and be leaders. These potentials and expertise can be utilised for school capacitation and improvement.

For this study the researcher wants to know about the resources for technological development of PLCs during the pandemic era, because semi-rural schools lacked computer labs and internet coverage. Many schools have no access to official websites, or e-mails to communicate with peers, parents and learners (Alhouti, 2020). Therefore, the COVID-19 pandemic challenged

almost everybody, even the well- groomed and knowledgeable teachers (Darling-Hammond & Hyler, 2020). The plan and substance of training programmes should be contain the required to be transferred to the school situation.

2.5.4 Professional evaluation reflective and feedback

Professional reflective ensures effective implementation of the subject-teaching methods. It is used as the platform to share ideas, concerns, successes, failures, uncertainties and opportunities. This platform will help the attitude of teachers and help them speak freely about our concerns and try to avoid the same mistakes in future. With this platform teachers are able to take risks and attempt online teaching, including several new teaching methods and skills.

Evaluation and reflection during this period of COVID-19 must be continuous and be a process to 'force' teacher leaders to take risks with unfamiliar gadgets and at times unfamiliar platforms to check which one will work best for teaching and learning. We have been under pressure, so we discuss issues that face us, think about solutions, and analyse our practice to improve online teaching quality. 'Each one, teach one' phrase on mentoring novice teachers during time can be productive for PLCs. This collaboration brings a smart transition, traditional teachers teach the novice teachers the traditional ways of teaching whilst novice teachers bring their technology expertise on board. Evaluation and reflection also happened within small groups motivated with friendship or common interest. The mentors might not be well-trained and motivated leaders when it comes to technology but developing PLCs with limited available resources will be a major asset to school leadership and teacher leaders in semi-rural schools.

In the interest of the study this can bring proper development on PLCs and best transition and massive improvement to online teaching and learning. The challenges that are met will be easily tackled. Important, and incurred, school improvement is likely to require school leaders to redefine their role as professional leaders, with a central focus on leadership for teaching and teach (Bush et al., 2010).

2.6 Barriers to developing and leveraging Professional Learning Communities

There are two main barriers to developing and leveraging professional learning communities and these are the lack of adequate support from departmental structures; lack of resources for the successful implementation of online education and resistance to change.

2.6.1 Lack of adequate support from departmental structures; lack of resources for the successful implementation of online education

Successful online collaboration in schools is faced with various types of challenges (Hew & Brush, 2007), especially in developing countries in the Sub-Saharan African (Aksal & Gazi, 2015). The lists of things listed by Sarkar (2012) were to be considered when implementing online collaboration learning includes; leadership issues, equitable distribution of resources and sustainability, and financial issues. Some of these challenges include district officials who have not succeeded in building collegial relationships between school leaders, teacher leaders and teachers. The district officials in many instances have fully committed to the implementation of online programmes prior to its commencement (Powell, & Border, 2019). For example, the introductory sections of the initiatives to improve the facilitation of technology-driven learning by the government have not been fruitful since the urgency of returning back to school after the COVID-19 closure was also rushed in South Africa (Meringue & Garidzirai, (2021). Engagement in the professional development process, which included planning to meet the needs of teachers and evaluating the effectiveness of the professional development, have not occupied in schools. All this is attributed to the lethargy of the district officials (Devlin & Fisher, 2021). It is reported in the literature that many of schools were dissatisfied with the way authorities operated (Devlin & Fisher, 2021). There have been challenges on the ground that had made school leaders to feel that they were not supported by the government. The officials in districts have not consulted widely as well. In short, poor interrelationships between the district officials and school-based stakeholders have exacerbated the problem even further.

One of the government challenges is money. This leads to a lack of materials for digital learning like smartphones, laptops, and so forth. Limitation of funds affected the quality of professional development to be brought to schools by the government. And this can be combined with the lack of expertise in the administration of various school programmes (Baldwin & Wilder, 2014). The other challenges include insufficient planning time, and unpredictable daily schedules for implementation whereas for organisational and leadership interventions to

succeed, initiatives like success of online learning should be implemented with minimum hustles (Aarons, Ehrhart, Farahnak & Hurlburt, 2015; Glisson & Schoenwald, 2005). In the context of South Africa, government policies have not been uniform applied across all the races. This is one of the challenges in schools during COVID-19 period, and the responses to this pandemic have tended to reflect racial disparities that are associated with the countries historical legacy of inequalities (Bhengu, 2021).

2.6.2 Resistance to change

The second barrier is about resistance to change. With COVID-19 pandemic, it has become clearer that education system is open to external dangers (Bozkurt & Sharma, 2020). Ribeiro (2020) rightly noted that this digital transformation of teaching and learning delivery came with several logistical challenges and attitudinal modifications. Resistance to change is embedded in deficiency and unhappiness, and it must be understood to teacher leaders who try to influence the instituted habits of their associates that there is emotional attachment and that teachers feel toward their well-known practices (Donaldson et al., 2008; Evans, 1996; Katzenmeyer & Moller, 2009). This can happen when teachers pull back from disagreements, shutting all the communication avenues, or show any sign of resistance that will lead to ineffective and inoperable PLCs (Donaldson et al., 2008; Evans, 1996). Katzenmeyer and Moller, (2009) cautions that at times, leaders can be wrapped up on advancement can become "swept up in the urgency of a problem and the promise of a solution, turned to neglect and misjudge the effort and distress of the individuals who must adjust". If the approach is systematic, there will be no last minute organising and known what is expected of individuals, I guess resistance can be eliminated during the pandemic era.

Some leaders lack the feeling of agency, and this can bring the schools downfall, because there are people who look up to school leaders and when they face challenges technologically, they can turn on them for support (Muñoz et al., 2016). When teacher leaders support their colleagues and teams in almost everything that has to do with technology develops their agency and sense of efficacy in technology (Holden & Rada, 2011). School leadership, especially the principal can be a stumbling block when it comes to change. If they do not engage and lack of communication to improve school culture and climate (Koonce et al., 2019), planning and organisation between teacher leaders and school leadership exacerbates problems for professional development process.

2.7 Strategies to mitigate barriers encountered by teacher leaders when developing and leveraging Professional Learning Communities

In this section, I discuss four strategies that the literature has identified as critical to mitigating the barriers to developing and leveraging the PLCs. These strategies are, generosity, appreciation and compassion; Setting sensible expectations for colleagues and having tolerance for other people and self; creating positive philosophy to produce professional abilities, capabilities and skills to enhance and broaden other leaders' expertise/cultivating and growing other leaders, and teacher leadership as the web of teacher collaboration.

2.7.1 Generosity, appreciation and compassion

Providing individuals with the opportunity of shared leadership contributes to a positive school culture which can result in increases in the learners' achievement (Leithwood et al., 2008). When principals enact core collaborative leadership practices, they can influence the motivation and beliefs which lead to significant influence on classroom practices of teaching and learning (Leithwood et al., 2008). Gilligan (2011) expresses that what is needed during unprecedented crises is an "ethics of care" because it is inductive, contextual and psychological, rather than deductive. School leaders, teacher leaders, teachers, learners and everyone in the education fraternity was during this time emotionally, spiritually and physically drained and demoralised by the COVID-19. All that was needed was a little bit of generosity towards one another in order to be able to cope with what was happening around us. This study looks deeply at what teacher leaders do in applying generosity of development of PLC during the confusing times. At times they themselves can be confused but they have to be ten times as generous as they can to others in order to move forward (Pe Narroja et. al. 2013). Gilligan (2011) further states that the world complexity is "like a puzzle with humans" when we look at it through the lens of "an ethics of care". As much as teacher leaders are not state paid leaders but because of their experience in leading at times they offer personalised kindness to their groups and are prepared to "go beyond the call of duty" in transmitting their responsibilities. Gilligan (2011) defines such leaders as caring leaders. Care, after all, is not just about the self; it is about active involvement with others (Ciulla, 2009). According to Gabriel (2015), the criteria to judge leaders is a fantasy and myth together with early life experiences. In the case

of COVID-19 in South African, schools teacher leaders were not trained to lead in such a time but they were expected to push as much as they can to keep teaching and learning on board.

COVID-19 pandemic was more of the national disaster, and thus political intervention was expected. Drawing from an ethics of care and compassion, Tomkins (2020) questioned Boris Johnsons' absence from handling the pandemic in the context of the United Kingdom politics. This scholar shares similar views as Ciulla (2009) emphasising that "the role of a leader includes caring for others or taking responsibility for them". At times schools and teacher leaders might not have all the resources and the technological skills to mitigate the barriers encountered but their presence is crucial to accomplish both the PLCs' development and leveraging, and to show that they care (Tomkins, 2020). Lowrie, (2020) perceives such leaders as not careless but rather carefree leaders.

2.7.2 Setting sensible expectations for colleagues and having tolerance for other people and self

Leaders who build a shared vision foster the acceptance of group goals and maintain high-performance expectations, create the conditions for establishing a collaborative community of teachers, learners, and stakeholders (DeWitt, 2016; Leithwood, Harris & Hopkins, 2008). The commitment and collaborative leadership have the ability to positively manipulate victory for all the learners. By incorporating professional development of teachers and leaders in the remedial efforts enhances learning and sustenance of the clients' wellbeing and integrates them with the standard professional development programmes. Teacher leaders reinforce relations between schools and communities to facilitate change and follow-up.

Successful teacher leaders tend to have a long-term view of education, and they have the skills to bring a school community together to establish an agreed upon direction (Fullan & Hargreaves 2016). They are able to articulate a vision for many years, and make sense of this vision so that school communities do not only understand what is happening in the present, but also understand how this vision fits with the future progress of the school. These leaders are people-centred, and are particularly focused on developing the staff (Yin et al. 2019). They are good at leading change and putting in place the organisational aspects that will lead to sustained success. Successful teacher leaders know about good curriculum, pedagogy and assessment, and how to help improve teaching and learning. To these four areas of practice there are at least

three other areas that help to promote success (Zahroh, 2015)). Successful teacher leaders understand that ultimately they are responsible for their own professional development, and they are proactive in their development and are restless for new ideas. They also understand that leadership is about influencing the behaviours of others in a deliberate process that leads to behaviour change. Finally, they understand the multiple contexts in which their school exists, and they are able to respond to, and often influence, these contexts (Kali, 2020; Lavonen, 2017). They become storytellers and sense makers to help others understand the place of a school in a complicated set of contexts.

2.7.3 Creating positive philosophy to produce professional abilities, capabilities and skills to enhance and broaden other leaders' expertise/Cultivating and growing other leaders

The discussion on the notion of creating a positive philosophy to produce professional abilities, capabilities and skills is drawn from the work of (Bobley, & Best, 2021). For a school to be viable, it needs to have an unmistakable, aggregate commitment alongside committed sustenance leadership at the school and district level. At the point when uncertainty, wavering and frailty struck, they would all be able to stand and work as the collective. School leaders and teacher leaders should analyse how school components are important for their COVID-19 reaction approach, they ought to ask: (a) WHAT is expected to help incredible teaching and learning?, (b) WHO is expected to realise a responsive and powerful community school procedure to help student achievement?, and (c) HOW to assemble a solid association utilising the methods and implementations?

District, school leaders and teacher leaders need to be innovative in providing sufficient cost-effective training and resources (Bardesi, Al-Mashaikhi, Basahel, & Yamin, 2021). Face-to-face contacts might be affected but with the help of technology teaching and learning has to continue. Some schools in South Africa ensured the continuity of learning while schools were closed, by utilising numerous remote learning channels, namely television, radio, online platforms, and home correspondences (Dreesen, Akseer, Brossard, Dewan, Giraldo, Kamei, & Ortiz, 2020). Furthermore, the government and the education sector began planning accelerated learning and recovery programmes to recuperate learning lost due to COVID-19 interruptions (Nugroho & Atmojo, 2020). Developing and sustaining PLCs, equipping teachers and learners with technological skills and supplying schools with needed resources will produce good learner outcomes. Teacher and learners must be properly skilled in order to cope with

conditions such as those posed by COVID-19 pandemic. The education sector, the schools' district and relevant stakeholders must collaborate to design new systems and school improvement strategies (Holmén, Adawi & Holmberg, 2021). Thus, this study examines teacher leaders together with their PLCs' mandates during COVID-19.

Bearing in mind that the schools in the study are from semi-rural, densely populated area, high-poverty district with limited resources in Pinetown, efforts that are geared towards helping the schools are desperately needed. Distributing leadership has the potential to builds capacity supports change and can increases the extent of change in leading educational improvement attempts (York-Barr & Duke, 2004). Distributed leadership needs to be connected with practitioners and it places emphasis on interactions rather than actions of school leaders (Harris, 2013). The sharing of leadership responsibilities develops interdependence among members and how one's behaviour impacts the organisation increases their contribution in decision-making (Fullan, 2001). Distributed leadership develops effective leaders who understand their roles and how it impacts the role others (Elmore, 2002). Daily challenges diminish when responsibilities are shared. When different leaders lead various aspects of leadership, school leadership become more productive in completing all of the tasks. Distributed leadership is a sign of transformation in leadership style. It transforms and raises motivation of both formal and informal leaders, as well as members to achieve more than what is expected.

2.7.4 Teacher leadership as the web of teacher collaboration

Teacher leaders are a vital segment of school progress and constitute a crucial factor of distributed leadership (Blose, Mkhize, Ngidi & Myende, 2022). Teacher leaders are influential teachers and influence schools' direction or procedure, have become an increasingly acknowledged device for transformation (Cambum, Rowan, and & Taylor, 2003; Stein, Macaluso & Nevins, 2016). Attempts to expand and improve the role of teacher leaders in steering instructional adjustment have become prevalent (Berg, Carver & Mangin, 2014). In this study, school and teacher leaders, teachers and learners themselves need transformation for a smooth transition to take place from traditional ways of teaching to digital/online teaching and learners. South Africa might not have the best resources that the education system needs for transition to happen swiftly but teacher leaders and teachers' attitude can make a big difference (Bhengu & Blose, 2022). Generating and sharing teacher technological expertise can be used to empower school and teacher leaders and teachers can build beneficial and joint

interactions. Discovering diverse expertise from teacher leaders can be productive in classrooms.

Brewer et al. (2018) found that teacher leaders can support change in others through a feeling of workforce, which can propel deliberate and cooperative practice in education. For teachers to be proficient in the online teaching, extensive and generous technology training was needed. Both school leaders and teacher leaders are supposed to work around the clock for the successful implementation of online teaching and learning in schools (Adarkwah, 2021). The study reported here indicates that through the use of PLC developments and teacher leaders' involvements has a potential to address the challenges brought about by the emergence of COVID-19 pandemic. Nowadays, teacher leaders are known as classroom teachers who allocate their expertise in countless ways (Nappi, 2014). In concurrence, Danielson (2006) states that teacher leaders' influence goes beyond their own classrooms because of the improvement in the quality of teaching and learning demonstrated by teachers, thus they are perceived as a set of skills. Teacher leaders are not confined to specific roles; they perform different roles in the organisations. The leadership network perspective is cultivated in the concept of distributed leadership (Spillane, Halverson & Diamond, 2001), which is influenced by social and contextual factors

2.8 Theoretical framework

The distributive leadership concept was created in late 2000s based on Lave and Wenger's (1991) communities of practice theory and from the speculations of distributive cognisance and action hypothesis to expressly zero in consideration on how initiative was supported as an action and extended across the collective and conditional settings (Spillane, 2001). This leadership relies upon shift in concentration in leaders' practices towards the entire organisational objective (Shava et. al., 2018). Distributive leadership implies appointing the roles and responsibility at numerous sources in the foundation levels and gives direction to esteem expansion (Harris, 2004). Distributive leaders centre primarily around creating collaboration effort with a common perspective and aggregate duty regardless of role or position (Keppell, O'Dwyer, Lyon & Childs, 2010) and build up a culture that advances commitment, adaptability and development based on shared trust (Brown, 2004) to achieve

change in an educational framework while operating with a collegial outlook (Bakshi, 2019). Teacher leaders influence the abilities and proficiencies of the colleagues by connecting distributive leadership method that would empower collaborative efforts and permits every staff to function uncompelled. Distributed leadership embrace and promote individual agency and collaboration rather than waiting for people in positions. Teacher leaders may apply a distributional leadership approach when assisting teachers during a crisis situation (Bolden, 2011; Murphy et al., 2009; Spillane et al., 2015), and during technology implementation. Distributed leadership is regarded as a democratic style of leadership that is relevant for supporting interventions and collaboration of community members (Shepherd-Jones & Salisbury-Glennon, 2018).

Different challenges are faced by teachers in trying to adjust to online teaching and learning and keeping the lines of communication with learners open learning, supporting and development. Nonetheless some teachers have mastered these challenges. Leaders who enable a supportive and open climate among team members contribute to the members' convergence around the team and increase bonding among team members, which are central aspects of group cohesion (Post, 2015). Furthermore a leader who fosters participative decision making increases employee's degree of job autonomy (Cheong et al., 2016, Mertens & Recker, 2020).

According to Fernandez and Shaw (2020), some of the best distributive leadership practices are; associating with individuals as people and building up common trust, distributing leadership all through the association and conveying plainly and frequently with all partners. Even if these prescribed procedures are illustrated here with regards to educational institutions rotating to online classes, confronted with the current untimely COVID-19 pandemic, they may similarly be implemented by practitioners working in different areas confronting their own crisis. Since interfacing with individuals and building up common confidence, just as transitioning to a common leadership concept can require some serious energy, those in leadership roles ought to focus on these duties promptly after tolerating the position instead waiting for another wave of crises to show up. The credits of compelling teacher leaders when confronting versatile difficulties have been recently depicted (Fernandez & Shaw, 2020); they are not restricted to issues of responsibility, reliability, and honesty.

2.9 Chapter summary

The review of literature presented in this chapter provided insights about teacher leadership role during COVID-19 in schools. Leadership theories and styles were discussed, including the challenges associated with their implementation. Strategies to mitigate the barriers encountered by teacher leaders when developing and leveraging PLCs, setting sensible expectations for colleagues and having tolerance for other people and self, and creating positive philosophy to produce professional abilities, capabilities and skills to enhance and broaden other leaders' expertise or cultivating and growing other leaders was also looked at. The chapter also presented distributive leadership theory, as it is the theory that embraces and promotes collaboration effort and permits every staff to function at most their optimum levels. The next chapter focuses on issues of research designs and methodology.

CHAPTER THREE

RESEARCH METHODOLOGY

3.1 Introduction

The previous chapter presented the literature review and the theoretical framework regarding the development of professional learning community turbulences such as during COVID-19 pandemic. This chapter presents the research design and methodology that was employed to generate data about the perspectives of teacher leaders in leveraging and mitigating the challenges encountered in the online teaching and learning as a response to COVID-19 pandemic imperatives. Bertram and Christian (2014) define methodology as the study of different methods through which knowledge is obtained and aims to present the work plan of the entire research. This chapter also discusses research techniques and steps that were applied in the process of generating and analysing data. The trustworthiness of the research and the

ethical considerations it entailed form part of the discussion in this chapter. Finally, a summary of the chapter is provided.

3.2 The research Paradigm

The term paradigm has been dissected differently by various scholars. A paradigm is a pattern comprising a number of assumptions about peoples' thinking or how one views the world. Research paradigm is described by Maree (2015) as a set of conventions and views on important aspects of reality that allows for specific views of the world. Bertram and Christiansen (2014) concur with this view by defining paradigm as framework that represents a particular worldview about what is acceptable to research and how this should be carried out. It is further defined as a conventional belief about what we think about reality and addresses fundamental assumptions taken on faith, such as beliefs about the nature of reality (ontology), the relationship between the knower and the known (epistemology) and assumptions about methodologies (Creswell, 2014; Nieuwenhuis, 2007). Researchers (Bertram & Christiansen, 2014; Cohen, et al., 2011; Creswell, 2014) acknowledge that people have different philosophical orientations about the world and nature of research. In addition, a paradigm provides guidance and directs thinking and action in relation to observed reality.

There is a vast number of paradigms as researchers view the world differently. The widely discussed paradigms are positivism; post positivism; constructivism/interpretivism; transformative/critical paradigm and pragmatic paradigms. This study is framed within the interpretive paradigm. By interviewing teacher leaders, I was able to glean information, which afforded me insights into understanding their daily experiences, both positive and negative, of leading to online teaching and learning in the midst of COVID-19. It also contributed valuable information in answering my research questions.

An interpretive paradigm involves methodological approaches that allow an opportunity for the concerns, voice and practices of the participants to be heard (Cole, 2006). In the context of this study, I ensured that she engaged and interacted with the participants and allowed them to fully express their views and experiences about the studied phenomenon. Punch and Oancea (2014) posit that reality is limited to context and cannot be generalised. The data from this study is not prone to generalisation due to the use of an interpretivist paradigm. Therefore, the

data that was generated will not be generalised into one common reality. Conversely, this paradigm had a great impact on how then I perceived reality and interpretations of the findings. Since this study was embedded on the interpretive paradigm, interpretivists assumed that reality has many truths; it was therefore imperative to listen to individual's social experiences.

3.3 Research approach

In this study a qualitative approach was used to as it satisfied the researchers' curiosity and desire for a better understanding of the phenomenon under the gaze. A qualitative research approach focuses on methods of investigation that consider the participants' culture, history, emotional lives and interactive activities (Berg & Lune, 2014). Qualitative approach does not generalise the findings from the generated data but it is more of a compass to find direction that one follows to uncover phenomena and helps in finding valid answers for research questions (Kumar 2011). Data is naturally generated from the participant's own setting. In this study, it was the role of teacher leaders in developing and leveraging professional learning communities during and post COVID-19 pandemic in schools.

According to Punch and Oancea (2014), a research approach is a fragmented fundamental plan which includes four key ideas; strategy, conceptual framework, the question of who or what to be studied, and lastly a tool design. One of the strategies is that qualitative researchers lean towards generating data in the participant's place. In qualitative research, the researchers do not invite participants in a neutral place and they do not send out instruments to generate data. They go to the participants' natural setting and directly interact with the participants to see their actions and behaviours within their own context (Creswell, 2014). Amongst other factors qualitative researchers are determined to understand social phenomena from the view points of the participants; sometimes they go beyond and generate a theory. Therefore, qualitative research approach was found suitable for this study because it allowed me to know participants personally, to understand their daily challenges and how they face these. This research approach assisted me in making interpretations and best described the participants' actions (Brynard, Hanekom & Brynard, 2014). It provides a unique example of real people in real situations, enabling readers to understand ideas more clearly than by simply presenting them with abstract theories or principles (Cohen et al., 2011).

3.4 Research design

This study adopted a case study as its research design. This study adopted case study design as it has the potential to achieve an in depth understanding of the research phenomenon. A case study design is described as a rigorous exploration approach that allows researchers to search and discover a researched phenomenon of the study (Flick, 2018), meaning a certain level of due diligence must be upheld. Flick (2018) further supports the use of a case study as it brings simplicity to the exploration of the research phenomenon. This view is in line with the aims of this study as it aims at gaining an insight on the exploration of how teacher leaders develop and leverage professional learning communities during and beyond the unprecedented time of COVID-19. The interactions with teacher leaders help to probe deeply and to analyse vehemently the multi-truths constituted in the phenomena (Cohen et al., 2011). In studies that seek the 'what is it like', case studies are appropriate as they allow the researcher to be up close and take advantage of the relations case studies allow to gain thick descriptions. Case study allowed me as a researcher to be exposed to the real context of the research phenomenon and behaviours of teacher leaders and situation that were being researched. A case study provided insightful learning about teacher leaders' leadership practices in semi-rural schools.

3.5 Selection of participants

When selecting my participants, I had to make certain that I have participants who want to participate in my study, and on the other hand, I had to make sure that I had participants that were going to provide a rich diversity of data about developing and leveraging on PLCs in times of crisis. The sampling methods that was apt for my study was purposive and convenient. Purposive sampling allows the researcher to select participants that are relevant for the study (Braun & Clarke, 2013; Rule & John, 2011). Purposive sampling can also be comprehended as subjective, judgemental and selective. All participants of this study were selected based on their geographical accessibility and willingness to participate in the study. Therefore, participants were only made-up of teacher leaders as I was specifically looking for their perspectives. As such, I intentionally selected four teacher leaders of schools located in semi-rural area. The following criteria were used to select participants for this study: (a) they must be teacher leaders during the COVID-19 era; (b) they must be under Pinetown District; (c) they

must be members of PLCs in their respective schools. Convenient sampling involved selecting participants that are accessible and convenient for the researcher. Purposive and convenient sampling methods are often used in small-scale qualitative studies (Cohen et al., 2011). My inquiry was a small-scale study.

3.6 Data generation methods

Bertram and Christiansen (2014) describe data as evidence that researchers gather to find answers to the specific questions they are asking and data is used to gather information for a certain study (Cohen et al., 2011). The researcher in qualitative inquiry is a vital instrument of generating information in a relational atmosphere. The concept of a researcher being a research instrument arises from the fact that the researcher personally goes and interacts with the participants in their regular venue (Cohen, Manion & Morrison, 2018; Henning, van Rensburg & Smit, 2004). The researcher does not control or manipulate the behaviour of the participants; hence, qualitative research is described as naturalistic (Cohen et al., 2018). This study has embraced the same rationale. I used semi-structured interviews to generate data. Semistructured interviews allow flexibility in that the interviewer can adjust the sequence on how topics are covered (Bernard et al., 2016). Semi-structured interviews were chosen because they allowed follow-up and probing, as well as clarifying some questions (Bertram & Christiansen, 2014). Cook (2008) and Kvale (1996) conclude that the semi-structured interview is neither a structured "questionnaire" with preconceived questions nor an unstructured conversation with no prescribed topics. It is a conversation where the interviewee makes account of his subjective experiences related to the theme set before him by the researcher, and the interviewer attempts to explore these experiences for further details worth analysing.

I had direct interaction individually with selected teacher leaders leading PLCs during the times of the pandemic to get their intimate information about their thoughts and heart-felt feelings about leading in under resourced schools during these trying times. I interviewed four teacher leaders, two from the primary school and two from schools located in the Pinetown District. My interviews took about forty to sixty minutes, sometimes five to ten minutes more because of laughs, giggles, anger and surprises. The interviews were face-to-face interviews and the COVID-19 protocols were highly observed, which was wearing of masks, using sanitisers and social distancing. The beauty of adopting qualitative, semi-structured interviews is that they are flexible and can be adjusted accordingly as the need arises (Cohen et al., 2011).

3.7 Data analysis

Analysing qualitative data is founded on some of key worldviews underpinning the study. In other words, the fact that this study was located within an interpretative research paradigm implies that the analysis technique has to align to this paradigm (Maree, 2015). Qualitative data focuses on smaller number of participants as compared to quantitative data which consist of large samples and huge quantities of data. Data was generated from four teacher leaders from two schools located in semi-rural areas. This means that the volume of data was manageable. This assisted me, as a researcher to analyse the data thoroughly and achieve detailed and rich data. Qualitative data analysis incorporates placing data in a logical sequence in terms of participants' descriptions of situations, observing patterns, themes, categories and regularities (Cohen et al., 2011).

The data for this study were analysed using thematic analysis, which is a process of identifying patterns or themes within qualitative data (Braun & Clarke, 2013). Raw data that was generated was the starting point; thereafter, patterns were detected (Bertram & Christiansen, 2014). All interviews were audio recorded with the participants' permission and were transcribed verbatim. This assisted with descriptive codes for analysis which aimed at identifying, linking and labelling the interviews to determine themes and patterns. The codes were then divided into categories that were evident in the data. The categories that were distinctive were grouped into themes (Creswell, 2009; Saldana, 2009). These themes were used as subheadings to group the findings of this study. This assisted me as the researcher in answering the research questions.

3.8 Trustworthiness

Norman (2020) infers that trustworthiness as "The degree of trust one has in the person telling the tale has much to do with the degree of trust attributed to the telling". Lincoln and Guba (1985) talks about four factors that are entailed in trustworthiness, which are credibility, transferability, dependability and confirmability. It is very essential that every study must be trustworthy by showing true picture of the phenomenon studied and participants' descriptions to enable the reader to decide if the situation is applicable other contexts (Korstjens & Moser,

2018). To ensure trustworthiness of the findings, I utilised these four factors, and are discussed below.

3.8.1 Credibility

Norman et.al (2020) says that the findings must be valid and believable in order for a study to be acceptable. Using different methods and source of information to generate data is one of the techniques of enhancing the study's credibility. However the used methods and sources repeatedly give similar outcomes. This can be done by using different methods and sources repeatedly, and they all give similar outcomes, credibility of the findings can be said to have been realised. This study was enriched by employing of digital recording device for data generation instead of writing down notes. I brought to the participants' notice that there was no wrong or right answer and that the researcher was looking for their point of view. In that way, the participants could speak freely in relation to what they had experienced. I also enhanced credibility by giving the participants a copy of the transcripts to listen to and confirm whether it was precise, if not corrections were made. Credibility of the research must reflect the reality and lived experiences of the participants (Bertram & Christiansen, 2014).

3.8.2 Transferability

The concept of transferability refers to a situation where the findings of a study can be replicable in another similar situation. Bertram and Christiansen (2014) state that data generated in a study can be applicable to the other study with the same context. To enhance transferability, qualitative researchers have to provide thick details about every aspect of the study, including the study context (Cohen, Manion & Morrison, 2018). Looking at the features of the qualitative study, the findings may not be generalised (Creswell, 2014). In qualitative research the findings are not generalisable. Proper data needs to be generated to support the investigated phenomena (Petty, Oliver, Thomson, & Graham, 2012). To enhance transferability in this study, I ensured that I give a detailed description of the research sites and also all the whole process of conducting the research.

3.8.3 Dependability

Dependability in a study is seen by being reliable and the distinctiveness to specific times and place. However, the concept of reliability is frowned at in qualitative research because reliability is a quantitative concept that is used in quantitative research that also comprises big samples for generalisability purposes. Therefore, reliability is qualitative studies is unattainable therefore to achieve dependability in a qualitative study. Credibility cannot exist without the presence of dependability, and credibility is truly the root of quality (Lincoln & Guba, 1985). To ensure dependability I gave a thorough description of how data was generated and analysed. Bertram and Christiansen (2014) substantiate that dependability is when the researcher can justify why there may be variations. In order to address the dependability issue more directly, they argue that the processes within the study should be reported in detail, thereby enabling a future researcher to repeat the work, if not necessarily to gain the same results in the study.

3.8.4 Confirmability

Confirmability is getting close to objective reality as qualitative research can get (Lincoln & Guba, 1985). Confirmability is about getting closer to the objectivity ensuring that the understandings and discoveries are from the participants' lived encounters and do exclude the researcher's biases. To achieve confirmability in this study, I used a number of techniques. The first one was to ensure that there is no imposition of my personal interpretation in the findings or the discussions thereof. The second technique entailed giving the participants the transcripts to check the accuracy of the interview recordings. To moderate these issues, participants must recognise their voices and confirm that the transcripts are a true reflection of the content of our discussion during the interviews (Korstjens & Moser, 2018).

3.9 Ethical Issues

Ethical consideration is the main issue in a research project, more especially because it pertains to human beings (Cohen et al., 2013). Lynch (2012) calls it as the ethics of the profession where professionals are constantly aware of their code of ethics, of what is anticipated from them and what they are relied upon to do to serve the well-being of the students. Participants must be educated that the study is wilfully and voluntary and that they can pull out whenever they fill like without any cause or notice. In short, the participants have to know their rights and

autonomy; and therefore, that they cannot be forced to participate in the study if they do not want to.

This study complies with the procedures of the Department of Basic Education in South Africa. The University of KwaZulu-Natal ethical protocol in conducting research was followed. The University gave consent and the Department of Basic Education in the province of KwaZulu-Natal gave me permission to conduct the study in its schools during the agreed upon period. The Human and Social sciences Research Ethics Committee (HSSREC) of the University of KwaZulu-Natal issued the ethical approval letter, and it is attached in the dissertation. I sought the permission of the four teacher leaders to conduct interviews with them on the research topic. The necessary documentation and permission letters were signed to indicate that the participants voluntarily consented to the interviews. I assured the participants that whatever they told me would be treated with the greatest confidentiality and privacy.

All the participants were furnished with informed and educating consent. During data generation process, every conceivable exertion was made to identify nonverbal signs of inconvenience or any sign of a wish to pull out from the interviews. All the participants were guaranteed and given assurance that their identity would be kept confident. Participants' personal information would not be exposed even under any circumstances. For identification of participants for research, pseudonyms were used instead of their real names. In advance, appointments and interview sessions were scheduled to ensure the timeous completion of the interviews. Participants agreed to their voices being tape-recorded. All transcriptions would be kept in a secure place to ensure strict confidentiality. All transcriptions were read by the participants to ensure that they felt comfortable with the information they imparted and if they wanted anything removed, it was accordingly done as they needed to be at ease with their responses.

3.10 Chapter Summary

This chapter deals with the research design and methodology. It includes the data generation methods or techniques used in the study. Research paradigm, research design and research approach of the study are explained as well. Sampling procedure and after the discussion of the analysis techniques used, measures to ensure trustworthiness are highlighted as well. Lastly, ethical issues are discussed. Chapter Four will present a presentation and analysis of research

findings about the research topic, which is about understanding issues around the role of teacher leaders in the development of professional learning communities during COVID-19 and beyond.

CHAPTER FOUR

FINDINGS AND DISCUSSION OF DATA

4.1 Introduction

The previous chapter outlined the research design and methodology that was employed in the study. This chapter, Chapter Four is the analysis presentation of research findings about the role of teacher leaders in developing and leveraging professional learning communities during and post the COVID-19 outbreak. This chapter analyses and presents the findings of the data generated from teacher leaders through the use of semi-structured interviews. The chapter discusses the following themes that emerged from the research data. These themes are as follows: (a) Participants' understanding of professional learning communities (b) Teacher leaders' role in developing Professional Learning Communities at the school (c) The impact of COVID-19 on the functioning of Professional Learning Communities in schools (d) Promoting cohesion and connecting opportunities for teacher learning in Professional Learning Communities during the pandemic (e) Challenges encountered in developing Professional Learning Communities during this time (f) Deprivatising practice and preventing teacher isolation with the COVID-19 protocols in place (g) The means or strategies tried to leverage Professional Learning Communities during the pandemic (h) Maintaining shared routines, (j)

The way forward for the Professional Learning Communities post the pandemic. The study's main research questions are:

- What are the teacher leaders' experiences of their role in developing and leveraging professional learning communities in their schools during and post the COVID-19 pandemic?
- What challenges do teacher leaders encounter in developing and leveraging Professional Learning Communities in their schools during and post the COVID-19 pandemic?
- What appropriate strategies do teacher leaders use to mitigate the challenges they encountered when developing and leveraging Professional Learning Communities in their schools during and post COVID-19 pandemic?

4.2 Profile of participants

This section provides a short profile of each participant. Pseudonyms were used to ensure participant anonymity, confidentiality, and ethical considerations. The study was conducted in two schools, one primary and one secondary school in the Pinetown District of the province of KwaZulu-Natal, South Africa. The two schools that participated in the study are from a semi-rural area. The data sources were teacher leaders.

Vesi, teacher leader at Kholwa Primary School

Vesi is a male teacher who is a senior teacher in the school. Vesi is a teacher leader at Kholwa Primary School. He has been teaching for thirty-four years. He has a Senior Primary Teachers Diploma (SPTD). Vesi is teaching English in Grade 5 and Grade 6. Like most primary school teachers Vesi has taught Afrikaans, History and Geography and Mathematical Literacy, Mathematics and Mathematical Sciences (MLMMS) in Grade 5. Vesi is a post level one educator, because of the years of experience, he is now a senior teacher and he sits in some of the School Management Team meetings.

Nomkhu, teacher leader at Kholwa Primary School

Nomkhu is a female teacher leader at Kholwa Primary School with thirteen years teaching experience. Nomkhu has a B.Ed. Honours. She is teaching Maths and Economics and Management Sciences (EMS) in Grade seven.

Gosakazi, teacher leader at Thandaza Secondary School

Gosakazi, a female secondary school teacher, and has nine years of teaching experience. Gosakazi is a teacher leader at Thandaza Secondary School. She has B.Ed. Honours degree with Maths and Science specialisation. She is teaching Mathematics in Grade 10. She has high knowledge of technology. She is technologically skilled.

Novangeli, teacher leader at Thandaza Secondary School

Novangeli, is a female secondary school teacher with 23 years teaching experience. Novangeli is a teacher leader at Thandaza Secondary School. She has an Advanced Certificate in Education (ACE), with specialisations in English and Biology. Novangeli is teaching Life Sciences in Grade 9.

4.3 Data generation and discussion

The data generated from the semi-structured interviews is organised into broad themes and subthemes. The themes and their subthemes are thoroughly discussed below.

4.3.1 Teacher leaders' understanding of professional learning communities

The teacher leaders in this study were asked the question: "What do you understand about professional learning communities (PLCs)?" This question assisted the researcher to find out if the participants understood or knew anything about what the PLC is. Evidence from the participants' responses show that they had varying understanding of what PLC is about. In airing their views Nomkhu seem to have an idea of what PLC is about while Vesi, and Gosakazi, did not seem to have a clear understanding, but tried to say something about it. The emphasis from these teacher leaders was that PLC is about collaborative learning where teachers share information, work together to face challenges, paves the way forward and develops each other. This was evident from the following excerpts of their responses:

Nomkhu:

I think the professional learning community is the group of teachers that can work together to create learning communities. PLCs help increase the effectiveness of education being offered to learners. I also think that professional learning community is a method of fostering collaborative learning among colleagues with a particular work environment.

Vesi

To me the term PLC is the same as subject committee; it is just a change of terminology. PLC is more or less like subject committees, where teachers are grouped together, plan teaching strategies and look at challenges in certain the subjects. They share information and assist each other in terms of subject matter which is very much required these days because of different kinds of teaching experiences, some are still novice teachers and some are experienced teachers. Therefore PLCs are for teachers to develop each other, as well as self-development.

Gosakazi:

It is a community; in this context, the community is a school, where there are teachers, learners and other people like cleaners, admin clerks, the school governing body and general workers that make teaching and learning possible. Not everybody in this community is a professional, I think when they talk about professional they are referring to teachers. In that case teachers learn from each other. If there are any developments, strategies and challenges happening in the school, each one teaches one. I think that is the professional learning community.

Novangeli declared that he knew nothing about PLC and complained about the Department of Education (DoE) that introduces new terms now and again till teachers lose track. When probed further, she associated the term to support groups or team teaching. Her response was:

Novangeli:

In all my teaching experience, I have never heard that word. As we are about to retire, new things are coming out. Almost every day in our teaching experience we hear and

learn new things. All these terms make the teaching profession not to be as interesting as before.... I think you are talking about what we call support group or team teaching.

The findings suggest that although the participants showed varying understanding of the concept of PLC, their explanations emphasised some key construct shared by the literature that it is about collaborative professional development where groups of teachers team up and support each other to enhance practice. In line with these findings, literature (Jones, 2010; Whitford & Smith, 2010; Yendol-Hoppey, 2010) corroborates the notions that PLCs are support groups of professionals that participate collectively in determining their own developmental courses, and to set up exercises that will drive their development. At the centre of the concept PLC, lays the reason of improving and developing learners' learning by developing teaching practice. Whitford and Smith (2010) describe PLC as small groups of teachers meeting regularly to engage in a systematic peer critique and support by sharing their own professional practices. Yendol-Hoppey (2010) further describes PLCs as teachers' shared commitment to student learning through collaborative practice and decision-making. PLCs often exist within grade or content level teams, but they do not have to be limited to one school. PLCs may provide benefits at many levels, most notably to develop learners' success (Jones, 2010).

4.3.2 Teacher leaders' role in developing Professional Learning Communities at the school

The participants were asked about their role as teacher leaders in developing PLCs at their schools. The four participants shared similar views about their roles as teacher leaders. The culture of collaboration is seen in the participants' responses that ensures and promotes effective teaching and learning. It is stated by the participants that teachers need to work together by sharing information and learning from each other. Teachers need to work with relevant material like ATPs. In response to what has been asked, they said:

Vesi:

My main role as a teacher leader is curriculum coverage. We develop each other by keeping track of annual teaching plan (ATP); whenever I meet with the PLCs. We check what needs to be done, and I try to assist by collecting as much information and share it in the best way I know how. So, my role is to lead, develop and to guide my team. In

some instances, we tackle challenges that relates to some subject matter. We try to cover what the ATP requires us to cover since ATPs are the order of the day.

Nomkhu:

My role as a teacher leader is to schedule and conduct PLC meetings, give support to teachers in whatever they do. I guide and supervise colleagues. If there is extra needed material, I give my all for that material to be accessible to the PLC.

Gosakazi:

To me, developing each other is a main issue. Being informative and relevant as a teacher leader is a must. Teacher leaders must be able to usher in new skills to PLCs. My role is to blend experienced teachers and novice teachers in order to be able to work together. As much as teaching and learning is about curriculum coverage but transformation is needed. My role is to monitor the process of teaching and learning together with teacher development. Self-confidence to novice teachers' is still a challenge. Since teachers know that we are approaching the new era in teaching, therefore development is essential for every school, teacher, subject and content. This will improve teaching and learning.

Novangeli:

My role is to plan and call my support groups meetings. We look at the curriculum and what it entails. I encourage and motivate my group members to work in one accord. I listen to PLCs' challenges, derive new strategies and implement. The planning, implementing, assessing one another and giving feedback of work done, leads to successful teaching and learning. When planning the processes of teaching and learning, appropriate teaching and learning methods must be in place.

Teacher leaders' responses revealed that they had good understandings of their roles in the PLCs. At times, the understanding of teacher leadership can be mistaken with matters of office and power, whilst it is more of skills development and carrying out duties and obligations by teacher leaders. As teacher leaders continue leading and developing others, leadership skills can be seen. Leadership includes both assigned leaders, who hold formal leadership positions within an organisation as well as emergent leaders, who do not hold formal leadership roles but are influential to the group or organisational members and of whom group members are

accepting of this influence (Northouse, 2019). Sharing the same sentiment, Kotter; (2001) states that teacher leadership can either are formal capacities of administration and management, or informal capacities of teacher leaders and coordinators. Leithwood; (2003) states that teacher leadership is something of nature, continues to develop, and impacts schools and learners. Teacher leaders must be influential people who have the best interest for leaners and his team and be willing to contribute to transformation of education and learners' success. They must not only support the subject committees but must be actively involved in all activities. Their involvement must not be of a superior but of a peer who sees to it that team members have their share of the knowledge. They must ensure that team members benefit from the group. When looking at the role of teacher leaders' scholars (Admiraal et al., 2019; Maloney & Konza, 2011; Printy, 2008) emphasised the role of teacher leader as creators and facilitators of teachers' participation in PLCs. Huijboom et al. (2021) mentioned three steering factors to PLC – leadership, collective autonomy, and facilitation of group dynamic processes.

4.3.3 The impact of COVID-19 on the functioning of the Professional Learning Communities in schools.

Harris and Jones, (2010) and Hord, (2004) stated that when schools are developed as PLCs, their ability to change and improve is to build. Development in schools occurs when there is interaction between individual skills, interpersonal interactions, and organisational structures (Mitchell & Sackney, 2001) that is, the improvement of collective skills. The participants were asked to explain: "how COVID-19 had impacted on the functioning of PLCs in their schools" The participants highlighted that they had a problem with the cooperation of the PLCs functioning, not only on PLC but the school at large is affected as well. In response this is what they said:

Vesi:

COVID-19 has just stepped in the whole world and everything has changed. So, everybody and everything in the education sector has to adapt into a new normal. Whatever we are doing now; is totally different from what we used to do. The seating arrangement in classes, school attendance and teaching itself, all have changed. So COVID-19 has affected the PLC a lot and it affects even the curriculum coverage and the subject delivery. We don't meet as often as we used to with the PLC members. Others have bailed out of the group.

Nomkhu:

As far the PLC is concerned, COVID-19 impacted negatively in it. Now, it is difficult to meet face to face. We hardly meet as PLC and in our meetings members are demoralised by the changes they had to adapt to. Two of our members once tested positive to COVID-19 and were seek and that brought a major blow to other PLC members?

Gosakazi:

School functioning is no more the same for teachers, learners and the PLC. The pace in which the school functions is stressful. PLC members have phobia in being behind closed doors that affects our PLC meetings. Our ways of communication with members seem to have a downfall. Hence teacher development is affected as well as the process of teaching and learning.

Novangeli:

I think the PLC is becoming inefficient due to the complaints by PLC members who say they are overloaded which bring high pressure in their regular class work. They said they are trying to normalise the abnormal ways of working and PLC adds another pressure. Their main goal in school is to teach learners not teachers. Let us first get used to this new way of doing things then we will come back to PLC. That is my main challenge for now.

One can tell that COVID-19 has impacted negatively in these schools because the data states that the four teacher leaders in both schools share the similar sentiment about the functioning of PLCs in their schools. Vesi mentioned that members have bailed out from the PLC. Nomkhu said members are demoralised. Gosakazi stated the downfall in communication amongst PLC members and Novangeli said that their PLC as inefficient because their main focus is on learners not PLC. On several occasions, teachers have positive attitudes towards PLC but they are not willing to collaborate if they feel there is restricted and limited autonomy (Al Shammari et al. 2020). Thornton and Cherrington (2014) highlighted the role of teacher leaders, taking note of that appropriate and upheld leadership was fundamental for moulding trust relations. Upgrading the limit of PLCs further develops teaching and learning, supporting collegial

relations; empowering teachers' thoughts, individual and aggregate proceeding with proficient turn of events, and advancing a school culture of trust (DeMathews, 2014). Because of high death-rate in their district, teachers do not trust each other and this has caused the collapse of the functioning of PLCs in two of the schools.

To probe further, I asked the participants to please comment on how teachers interact and collaborate with each other in PLCs during this time. I then asked them if they are able to foster an atmosphere of trust. In response to my questions, the participants said:

Vesi said:

It is very much difficult because as we speak of this pandemic, with the set meeting programmes it is only the certain individuals who will come to the meeting. They are scared of those who have tested positive COVID-19 among them, even if they are fine.

Nomkhu:

Members started fearing for their lives. Being around members that were once infected put their lives at risk therefore they prefer not to come to meetings which cause a major drawback on collaborative teaching and interaction.

Gosakazi:

The level of distrust among teachers has escalated. Collaborative teaching is not happening due to COVID-19. Teachers have opted to work as individuals. They have decided to live with the fact that 'what is mine is mine.' To cascade the information, is not as effective as when PLC meet face to face.

Novangeli:

Not interaction, no collaboration and no trust amongst members. Teachers do not want to come to meetings to address issues. Strengths and weaknesses that members have among are now known as we speak.

In both schools participants acknowledged that collaboration, interaction and trust have been negatively affected by the pandemic amongst teachers. Through collaboration and interaction problems that arise are easily solved because teachers draw inspiration from one another. Hord

(1997) advocates that when PLCs work together with teachers, problems are solved together. Teachers collaborate to discuss strategies to improve learner achievement. Hord (1997) argues that collective creativity helps teachers to learn and discuss new teaching issues in order to improve learner outcomes. The roles of leadership should arrange conditions for excellent education that will produce improvements in learner outcomes (Leithwood et al., 2019). These conditions will incorporate ideas from other leaders of supporting and training colleagues (Breitkopf, 2018). Teachers are not working as teams anymore; they prefer to work as individuals. Teachers are demotivated and are not committed; if they are not working as a team new ideas and knowledge are not contributed. Scholars such as DuFour (2011) and Sargent and Hannum (2009) agree that effective collaborative learning helps teachers to learn new ideas about teaching and about learners learning.

Three participants reported that their instructional activities as PLC had become ineffective and inefficient due to low level of cooperation among PLC members. Teacher leaders working with teachers on classroom observations and constructive feedback conversations are put on hold for a while because of the pandemic that is reported to be escalating. When participants were asked: Do you continue to monitor the work of PLCs and provide constructive feedback? They responded by saying the following:

Vesi:

It is very difficult because it is said that the levels of infections are escalating, and that affects teachers as well. Remember that when people have tested positive, they have to stay home at least 10 days and there is no recovery and monitoring plan in place. When the teacher recovers from the infection and come back to work a lot has been done on curriculum coverage, monitoring and feedback. So, obviously, s/he will be behind in of a terms' work. So, it is a challenge indeed.

Nomkhu:

Teaching and learning is a process. PLC is not only about developing each other effectively it is also about teacher sustenance in spite of hindrances and problems as well. On a small scale with those two or three members who are still on board, I monitor, give support and constructive feedback.

Gosakazi aired her dissatisfaction by saying:

Monitoring and giving feedbacks to PLC members has dropped. Lately, communication has become difficult among PLC members. Escalation of COVID-19 infection cases is causing a major problem in my school. But for the fact that we are still committed in teaching and learning, we are trying. There are those individuals that are still as positive about collaborative teaching. The procedure has not changed we plan, deliver, monitor and give feedback with those few individuals.

Novangeli:

In fact, the process monitoring, reflection and feedback has decreased during the pandemic. For instance, lack of adequate support in the process of teaching and learning is the cause. Bringing almost everyone on board is sometimes difficult because there are no open dialogues that are taking place in our respective PLC due to COVID-19.

The voices above stated clearly that communication and collaborative teaching in these two schools is voluntary to PLC members who still see PLC as the platform for teacher support. Vesi is concerned about the level of COVID-19 infections that affects teachers' and their performance. The three participants' responses stated that the level of monitoring and feedback has dropped tremendously in their PLCs. If there is cooperation within PLCs teacher leaders have time to monitor progress, give feedback and provide support to teachers and still be able to teach their classes. Teacher leaders know that teachers gain confidence and trust to members and feel that they are being valued in the PLC. Feedbacks enhance critical self-reflection whether it is done casually or systematic. When novice teachers are monitored and given feedback in their teaching practice plays a big role in fine-tuning their roles in the PLC and the school at large.

Instructional leadership theory perspective is that trust between the teacher leaders and the teachers will lead to open and flexible relationship that allow teacher leaders and the teachers to communicate from their sense of truthfulness (Wanzare, 2012). This increases the level of respect, understanding and accepting differences amongst colleagues at work. Communication builds trust between teachers and superiors and maintains confidentiality. Instructional leadership theory promotes proper collaboration and communication.

To probe further, I posed this question to them; Are you able to address the most pressing instructional challenges in your PLCs? And their responses are indicated below.

Vesi:

Fortunately, now, yes... When schools opened for the first-time last year. I had no idea of how and where to start but as time progresses things began unfolding, and I began to understand a thing or two about COVID-19 and things started to loosen up. So, it becames easier for me as the teacher leader to request teachers to share with me or the PLC their challenges they are facing. There are those challenges that can be tackled overnight and there are those ones that need more brains. But in one way or the other, we as the team are able to address challenges.

Nomkhu:

Yes, I'm able to address challenges. During this pandemic, PLC membership has decreased. As much as developing and sustaining PLC is a challenge on its own but 'the fewer the members the fewer the challenges'. With the few members left communicating has turned out to be much easy.

Gosakazi:

Due to COVID-19 problems, PLC transformation to access teaching and learning support had been scarce among teachers. Teachers have adopted the passive role when it comes meeting, sharing of ideas and information and to teaching practices. No interaction of any kind among them.

Novangeli:

If I have the power to bring teachers together to talk, plan and work together for the sake of learners I was going to, but since everyone here (PLC) has a mind of her/his own it is hard for me to do that. The pandemic has destroyed our wonderful profession.

Three participants communication between them as teacher leaders and PLC members is dented. Vesi and Nomkhu mentioned that they could not say that there are no challenges but, they are doing their best to work with the few remaining PLC members. On the other hand, Gosakazi and Novangeli mentioned that there is no communication and no interaction among PLC members in Thandaza Secondary School. The worldwide perspective of online education has been seen as a good-to-have alternative but in all honesty it was not a model to ensure consistence of instructional activities (Ribeiro, 2020). Other schools find it easy to transform from traditional ways of teaching whilst others the migration process leads them to crisis due

to the pandemic, as cited by Hodges et al. (2020) and Manfuso (2020). The crisis-response migration procedures focused on the development of instruction online that will accommodate flexible teaching and learning.

4.3.4 Promoting coherence and connecting opportunities for teacher learning in the Professional Learning Communities during this pandemic

The top-down traditional ways of professional development are often occasional and unclear at times to impact teachers. Many times teachers were given one very shot presentations by the experts. They were expected to implement without having acquired or demonstrated any understanding and monitoring skills. PLCs provide a better option for teacher development which is more peer centred. Even though it does not automatically create more coherence and connected opportunities but it is theirs and they are able to develop it in the way it suits best. Teacher leaders are responsible to encourage and motivate teachers to work together by sharing information.

In the context of this study, I posed the following question to the participants: "As a teacher leader, are you able to promote coherent and connected opportunities for teacher learning in your PLCs during this pandemic?", and in responding to the question this is what they said:

Vesi:

Stigmatisation is still one of the major challenges in promoting coherency and connection opportunities for teacher learning in PLC in my school. Teachers who have had COVID-19 before, most of the time, are labelled as such. When it is time for PLC meeting other teachers still find it hard to come to the meetings. Others prefer to come to me individually to ask about the PLC progress, others ask from other members who are still involved with PLC. Therefore, it is still hard to connect with others as one big unite group PLC.

Nomkhu:

Face to face meetings seems to be problematic because of COVID-19. If one suggests or calls for zoom meetings, it is like opening a can of worms because some colleagues are technologically challenged. They do not know how to use zoom; others will

complain about unstable network and others will complain about data. We do not have a clear, logical and systematic way of connecting.

Gosakazi:

Not exactly... not as before. We used to meet anytime, anyhow and when the need arises not worrying about social distancing. Now teachers find it difficult to ask whenever they encounter problems. They have the tendency of not wanting to be close to others. Clear meeting schedules and working programs that we had are now inactive. People seem not to be enthusiastic about them, not only meetings also about learning from each other as well.

Novangeli:

Encouragement of higher levels of dialogue should be done because teachers are now living in fear. They are fearful of everything and anything and when they are in that mode they retreat.

One can claim that there is no coherence or connection in the two schools. Teachers have taken a 'backseat' when it comes to working with others. Teachers are living in fear of being infected by COVID-19. Vesi mentioned that teachers who have been infected with COVID-19 are stigmatised, that collapses coherence. Nomkhu stated that even communicating through zoom is a problem because teachers are expected to be provided with data to use zoom. Gosakazi stated that teachers did not want to adhere to meeting schedules and working programmes. Novangeli is opting for open dialogue to calm fear among teachers.

Capra (1996) states that the theoretical level of connectivity are firmly identified with systems/complexity thinking where associations around the globe are made to adopt a powerful harmony between framework constituents at a specific time. When PLCs in schools can promote connectivity and coherence during these tough times they will achieve much in due time. As time goes on, framework can be broken due to diversity and individualism of certain people. Even if times change, PLCs can still stand together. The teaching practices through PLCs encompass increasing connectedness and interdependence between teachers and learners, the exchanging of teaching materials and sharing of information (Ghemawat, 2003). PLC members and teachers need to understand the importance of togetherness. Blömeke and

Delaney (2012) mention that connecting concepts, processes, and representations increase familiarity with basic constructions and ideas for learners and teachers. Hiebert and Carpenter (1992) recommend that it is vital to make connections if one intends to develop understanding, skills and knowledge and advance teaching and learning to support in different ways.

4.3.5 Challenges encountered in developing Professional Learning Communities during this time

New policies must be dispensed to help teachers and schools during these trying times. Developmental programmes and capacity building workshops must be the basic piece of invalidating the challenges teacher leaders and teachers experience when they are developing PLCs and managing teaching and learning in schools. Relying on different interested parties to share their expertise in solving the existing challenges in school is a working solution to the challenges of PLC and teacher leaders. The provision of the incentives for those who do well in different categories helps eliminating some challenges. I also asked the participants the following question. What challenges (if any) have you encountered in developing PLC during this time? Their responses were affirmative, and explained the types of challenges that they encountered. The following extracts illustrate their responses:

Vesi:

The belief of individual working to other teachers is a huge drawback to the PLC. These teachers are keeping their thoughts and views to themselves; they are not even willing to share with their PLC members. Others even mention that they have to much on their plates because a lot needs to be done in a short space of time for learners' progression at the end of the year. The frequent meetings that used to take place among PLCs are no more. That leads to dysfunctionality of the PLC.

Nomkhu:

We used to have objectives and goals to be met by our PLC. It seems that the level of commitment for members has dropped. That has caused my energy, enthusiasm, moral and love level as the teacher leader to decline due to the workload caused by trying to uplift members. The PLC has become dysfunctional. I have is proper participation in my regular class work.

Gosakazi:

Fortunately, about two or three years back an NGO donated laptops and overhead projectors to our school for both teachers and learners. Teachers did not want learn how to use the gadgets. Now with the pandemic at hand, nothing has change they still do not want to learn that is making matters worse because we are faced with this time that demands technological skills. This is shocking because teachers are reluctant to meet face to face and when the alternative is given they still do not want to involve themselves, they confidently say that technology is not for them. They have taught before without technology they are still going to teach even now. They do not have the right attitude towards technology/online teaching.

Novangeli:

Challenges are there and they will always be there because we are working with people and PLC members do not want to adhere to whatever the group is doing. The pandemic has caused other teachers not want to work as a team. It is not that before the pandemic things were excellent but the level has dropped drastically.

The participants emphasised that they were not happy about the performance level of PLC in their respective schools. Vesi and Novangeli saw individualism as a challenge that leads to dysfunctionality of PLC in their schools. Nomkhus' moral had declined because of the level of commitment for PLC members. On the contrary, for Gosakazi the school was provided with technological gadgets but teacher did not want to be technologically skilled.

Before the pandemic, online teaching training was not included in teacher development plans. Alhouti, (2020) maintains that when the Department of Education fails to provide financial help for public schools, the implementation of online learning fails as well. Existing studies by Maloney and Konza, (2011); Printy (2008) and Watson, (2014) provides evidence that sharing expertise and knowledge fosters growth and learning and solve teaching issues for teachers in PLCs and it also improve learners' learning and school performance (Ratts et al., 2015; Thompson et al., 2004).

4.3.6 How to 'deprivatise practice' and prevent teacher isolation with the COVID-19 protocol in place

This theme is mainly about going back to the ways of doing things in a collaborative manner, and move away from individualistic tendencies that may have developed due to isolation caused by the COVID-19 pandemic. Because of the pandemic, a number of teachers have become accustomed to having privacy and independence and to teach in a manner in which they want. Some of them do not even want others' contributions to their ways of teaching. Such behaviour has caused their colleagues not to be able to help where necessary. Curry, (2008) mentioned that working in groups like the PLCs helps teachers to share teaching practices and challenges and confidently reflect on their work. Isolation has caused teachers in this study to not return to the past discussions and learn from others' experiences. Teachers' hesitance to work together is the obstruction of their cooperative learning in the two schools. This has come out strongly when I posed the question; "With the COVID-19 protocols in place, how have you tried to 'deprivatise practice' and prevent teacher isolation?" Their respective responses are captured below as follows:

Vesi:

COVID-19 has its protocols in place like social distancing and sanitising or the constant washing of hands and of wearing of masks. With my understanding COVID-19 protocols were a strategy to minimise the spread of the pandemic in a school environment instead they are individualising teaching and learning processes. Proper workshops or training to educate teachers about the pandemic protocols were very minimal before the school reopen and when teachers return back to schools they were scared of being in close contact with anyone. The shortage of PPEs escalated the problem. Teachers now prefer to isolate them themselves from the group to avoid being infected. To avoid teacher isolation developed overlapping exercises to that will accommodate close and distant colleagues.

Nomkhu:

I suggested zoom meetings to the PLC members to prevent individual teaching and to avoid isolation. My suggestion was not warmly welcomed by the majority. The few that accepted the idea are not always present for zoom meetings but ideas and challenges are shared and new teaching strategies are developed. Zoom meetings will capacitate teachers and promotion of learner and school improvement in our individual spaces.

Gosakazi:

In fact, these protocols instead of helping individualism, it promotes isolation. Some of the teachers prefer to be on their own to be where they are comfortable. They do not go to staffrooms anymore. Greetings, small chats and sharing of family stories in corridors are gone. Providing a way forward for teaching and learning activities must suit each and every ones' personal needs in order to build the effective relationships and to ensure social sustenance.

Novangeli:

We are dealing with people; people will always be people. At times they resort to isolation because they are not keen to share their information and for other people to know their problems as well. They see PLC as a platform that is not for them because they are not eager give assistance to others. Sharing information with teacher leaders is fine with them but not someone else. Broadening the concept of care is not for them.

The introduction of COVID-19 protocols was meant for good outcomes but because people have different views it ended up being the one that promotes individualism and isolation in these two schools. Vesi, Gosakazi and Novangeli share similar sentiments about teachers view on deprivatisation of isolation. The participants aired their frustrations on COVID-19 protocols that brought high level of isolation. One example of this was Nomkhus' zoom suggestion that was rejected by the majority of PLC members. According to Levine et al. (2015) PLCs are not the same novice teacher's voices are not heard they are told "you're first year, don't say anything, just sit there and be quiet" which indirectly destroys their confidence.

Other teachers feel like they are losing authority and control when using technology to teach. Ertmer (1999) terms these feelings "the second order barriers" that include attitudes and resistance to teachers. In agreement with this notion Chigona et al. (2010) states that other teachers perceive technology as a threat and something that is not beneficial to teaching and learning. Teachers who have no experience in the online teaching or have difficulty in operating technology and information gadgets usually have problems and challenges in smoothness of instructional activities (Conrad & Donaldson, 2011; Ko & Rossen, 2017; Watson, 2020). Patrick et al. (2000) affirm that teacher enthusiasm is directed at achieving high teacher performance but the lack interaction physical leads to weakening of enthusiasm to teach (Hennessy et al., 2005; Ruthven et al., 2004; Strunc, 2020).

4.3.7 The means or strategies tried to leverage the Professional Learning Communities during the pandemic

The major part of teacher support is professional development. The majority of the participants concurred that development of staff was an essential element in uplifting the standard of teaching and learning. The most common cited by almost all four of them was the use of workshops, whereby different experts can be invited to the school to offer skills that most staff members or even for one teacher's need. When participants were asked the question: "What means or strategies have you tried to leverage PLCs during the pandemic?" their responses in which they described strategies they used, they highlighted the following:

Vesi:

Strategies to leverage PLC for now, I think, it is a far-fetched theory. When schools were opening during COVID-19 teachers who were expected to work whilst were not catered as far as teaching skills under the new normal is concerned. The government must now think of what to do in order for us to continue with smooth and effective teaching under this COVID-19. Digital schooling system must be introduced to schools today and relevant gadgets must be provided, and proper training to use the gadgets must be provided as well. This strategy can then be proficient to both teachers and learners.

Nomkhu:

As hard as it is to meet face to face and with zoom, we have a WhatsApp platform where we talk about everything pertaining teaching and learning. Teachers are engaged in continuous activities of our PLC. At all cost, we are trying to sustain one another's support and networking. It is not every member who participates but we work with those few who are present and willing to cooperate.

Gosakazi:

The on-going engagements about classroom management, content for teaching and learning, curriculum coverage, encountered challenges and outcomes in support for professional development are things to talk about in our PLC meetings. At times it is not easy because not everyone showed up but we push in spite of that.

Novangeli:

The majority of our traditional members did not want transformation to the proposed changes but because of changing times they find themselves on the fence. The introduction of new skills mastering of new teaching approaches that are learner-centred will ensure effective teaching process.

The data indicates that the participants concurred that traditional or face-to-face ways of holding meetings are slowly expiring. It is clear that they all see technology as one of the strategies that can be used to deal with the barriers that they had encountered to leverage the PLC especially during these trying times. They are driven by the fact that having a few that are receptive to change, the future is bright and they will win. They know that no matter how few they can be, but they are willing to push forward. Their main problem is the traditional members of PLC. The data revealed that some participants did not cooperate with others despite of the obvious benefits to do so.

Bozkurt and Sharma (2020) argue that it has become obvious that the education system is not exempted from being swayable to external threats with COVID-19 pandemic, Ribeiro (2020) adds by stating that digital transformation came with a number of planning challenges and defiance adjustments. To tackle the challenges that accompany the introduction of digital teaching as an emergency, Ala-Mutka et al. (2008) suggest that schools should not develop isolated programmes for learning digital skills; instead all the subjects should be inserted in teaching and learning processes of skills development. Omotayo and Haliru (2020) further state that motivating learners to be digital competent will let them to remain relevant and modern.

Before the arrival of COVID-19 pandemic many countries did not include online teaching and learning as part of teacher development programs. In agreement, Kaden, (2020) and Trust and Whalen, (2020) maintain that in many countries the imperatives brought about by pandemic protocols have brought to the fore more responsibilities for professional development and staff meetings. Harris et al. (2020) argue for the significance of a PLC during the pandemic for demonstrating the 'power of collective professional learning' in the absence of traditional learning. The participants were asked the question: "Have you tried using digital networks?" their responses are given below.

Vesi:

No, I have two reasons. Firstly, teachers do not have electronic gadgets to use when teaching they will need data and skills to use the gadgets. Secondly, the learners will

need to have gadgets as well for teaching and learning to take place. Our community has high rate of unemployment.

Nomkhu:

Not fully...We do not have the gadgets like laptops or computers to use for meetings, collaborative teaching and teacher development programs. We use WhatsApp for PLC meetings and other special announcements. At times I do not get full cooperation from teachers because they complain about data.

Gosakazi:

No. Digital networks are not the answer for now. The answer is the teachers' attitude towards technology. They are negative about learning new things yet the era we are facing forces the whole education sector to change.

Novangeli:

Not fully...Since we have two groups in our PLC, experienced and novice teachers. Experienced teachers lack skills to operate computers, laptops and other technological gadgets. They do not even want to learn how to use their smartphones as supplemental for online teaching and learning.

Two of the participants have said they have not tried using digital networks because of the so-called experienced teachers' attitude and the socio-economic status of the school community. The other two participants have partially used online communication through WhatsApp to cascade the information amongst PLC members. Prensky (2001) speaks about learners who are born and raised during digital period as 'digital natives' who are expected to be technologically competent and confident. Even though a number of them do not have expected skills (Bennett et al., 2008), that trigger Shariman et al. (2012) to say that properties of new digital technologies to outline knowledge is yet to be discovered completely.

Beddoes, Prusak and Barney (2019) as well as Beddoes, Prusak, and Hall, (2014), describe technology as one of the benefits of PLCs and a potential for teachers to shape their own destiny and overcome the barriers they encounter. Online teaching can be effective if teachers and learners can be sufficiently prepared and supported. Apparently online teaching and learning

might present additional benefits that could lead to better results, irrespective of technological and socio-economic issues faced by the marginalised social groupings that have experienced limited access to education. To probe this issue further, I posed the following question as: "What are the benefits and/or the challenges of using digital networks to leverage PLCs?" Their responses were as follow:

Vesi:

If digital networks are used by our PLC it would have been beneficial by influencing our style of teaching, proficiency and development. I think the department must move away from supplying schools with books, instead, the department must provide schools with the e-books in order for schools to move forward. Like I said, teachers themselves are not willing to try using technology maybe if teachers are provided with electronic gadget that can be the change of attitude. Most teachers have no computer skills and that causes them to have no confidence in digital networks.

Nomkhu:

Technology has been beneficial to our PLC; the WhatsApp platform that we have used has shown the commitment, genius, and creative side of those who are involved in keeping PLC alive in the school. Tremendous developments brought in by novice teachers have led us to the new normal. We have created a vibrant PLC that is ambitious and has hope for tomorrow.

Gosakazi:

If teachers can be positive about digital network, it can be beneficial. PLC members' commitment varies. Most members sing praises for digital networks at a distance, as something for members not them. Digital networks can be fruitful and productive and learners' results can be a successful if teachers are adequately supported. Therefore, implementation under these situations is still a challenge.

Novangeli:

Teachers need a broader perspective of digital networks and consciousness of working in a PLC during these trying times. Engagements where teachers can realise the importance of their growth and skills development that can be brought by technology during these times must be introduced. Teachers' empowerment and improvement is eminent for themselves, the learners and the education sector at large.

Three of the participants, Vesi, Gosakazi and Novangeli can attest that digital platforms are beneficial but have not used them because of their colleagues' attitude. Nomkhu and her PLC had improvised by using WhatsApp platforms just get somewhere instead of letting PLC collapse because of traditional teachers.

During these crucial times, schools have investigated time-productive and significant approaches to give high standards of professional development to teacher leaders that would uphold both their digital skills and give professional help from peers (Jin & Redish, 2020). Sadly, research about equipping teachers with skills for online teaching regularly includes restricted detachments of information and skills without a significant direction on how it must be taught. That is true for the two studies that include teacher leaders and teachers (Moore-Adams et al., 2016; Parrish & Sadera, 2019). It is recommended that consistently consolidating technology use can benefit the schools as well as the teachers (Cooper et al., 2020; Gosselin et al., 2016; Norton & Hathaway, 2015).

4.3.8 Maintaining shared routines and the way forward for the Professional Learning Communities post the COVID-19 pandemic.

Fullan (2001) states that leadership is not the sole responsibility of the SMT members and teacher leaders, instead leadership is for everyone who is willing to promote collaboration, democratic group processes and create relationships. Teachers control institutional methodology, educational and technological help, required time, teacher expertise, and learners' capacity and willingness to apply technology (Fullan, 2001). PLC members must at least take ownership of the group. Ownership will cause PLC members enthusiasm for teaching to increase and be motivated to develop new ideas moving forward. If PLC members are not enthusiastic about assembling goals and objectives but teacher leaders must be motivated to do so. In this regard, the participants were asked the question: "To what extent are you able to maintain shared routines in your PLCs?" Their responses were as follows:

Vesi:

COVID-19 has taught us that teaching and learning does not only take place in the school and within the four walls of the classroom; technology can help us in this issue. Teacher's participation has slowed down tremendously during COVID-19. Therefore successfully sharing PLC leadership responsibility routines is a challenge as well.

Nomkhu:

Maintaining the shared routines is a bit challenging at the moment. Since we are at the peak of COVID-19, colleagues' commitment is not guaranteed. As much as we are not meeting face to face and virtually but WhatsApp platform is helping us. When one is expected to take a certain role in the PLC and only to find out that that individual is not at work due to COVID-19 and the remaining members do not want to help and take responsibility with the tasks at hand. Individual lack of commitment and ownership of the PLC is still high.

Gosakazi:

It is not maintained in a way that is supposed to happen since PLC members are not as committed as before. Challenges of development are escalating. Sharing of ideas and expertise are compromised by the lack of commitment and dedication. Hence I say to you maintaining shared routines are happening at a very minimal scale.

Novangeli:

Teachers are challenged by this pandemic period and they bail out from any responsibility given to them whenever they feel like. Teachers have lost confidence in almost everything and everyone, the education sector, the schooling system, PLCs and even themselves.

From the above utterances I can tell that the maintenance of shared routines is not happening in the two schools (Thandaza Primary and Kholwa Secondary School). I can also claim that fear is the major challenge in these schools because teachers did not feel secured. There is a lot of tension among them. The future of their PLCs is uncertain.

During these times of the COVID-19 pandemic and beyond, strategic methods of teaching, adjusted educational schedules must be reviewed by the usual forms of teaching, and find

options that will motivate and promote teachers and learners' emotional welfare even at a distance (Couto et al., 2020). The lack of interactions between teachers and learners because of the effect of COVID-19 pandemic was the main challenge because it is imperative for traditional teachers to adapt to classes, resources and accomplishments of different version (Basilaia & Kvavadze, 2020). The participants further aired their views about the way forward for PLCs post the pandemic. The participants had this to say in that regard:

Vesi:

Teachers are overworking in a school that does not have resources to teach in these trying times. Teachers and learners must have computers, internet and proper training to use them because most of them are computer illiterate. What is imminent now is for our education system need to be hands-on.

Nomkhu:

There will be a way forward after pandemic COVID-19 because teachers always need each other's expertise, and they will always need support from teacher leaders and their peers. They need to work collaboratively to produce good learners' results. The success of teachers, learners and the success of their schools lays on how effective the PLCs are. Therefore, I think that there will be a way forward after this pandemic COVID-19.

Gosakazi:

I do not see any 'post COVID-19'. When is that? It is said that as from December 2021 onwards there is going to be the fourth wave? This pandemic is our new normal now; all we need to do is to adapt and learn to live with COVID-19 as part of us. Whatever we are planning from now onwards must be in-line with Covid-19 protocols. Really, I do not see any 'post pandemic' happening.

Novangeli:

In the education sector the way forward can be done through progressive trainings and workshops within the PLC members. Such workshops can provide opportunities for advanced teaching and learning which is done through peer coaching and study groups. The use of significant tools or proper gadgets will keep the education system afloat in the midst of the trying times.

On the question asked, the data shows that two of the participants Nomkhu and Novangeli are positive about the 'post pandemic era' if the Department of Basic Education can intervene to redeem the situation. Vesi is on the fence about the 'post pandemic era'. Gosakazi sees no 'post pandemic era' because of the predictions about uncertainties relating to the pandemic.

Beech (2020) likens the technological innovation opportunities brought by COVID-19 that ushered in by the rocket technology and digital computer. Monique Sendze as rightly quoted in Manfuso (2020), views Information Communication Technology (ICT) professionals responded quickly to the crisis to provide solutions, with the current technological interventions provided by ICT professionals during this COVID-19 pandemic. There is no doubt that they are up to the task in providing more if more crisis erupts. Thus, this can be adopted in continuity of business and adversity rescue strategies.

After the participants have aired their views about PLC and the present and post COVID-19, at the end of the session I posed the following question: Is there anything that you would like to add as your closing remarks? This question was not directed to any specific issue relating to COVID-19. The participants responded as follow:

Vesi:

Firstly, the education sector must move away from the mentality that teaching can only take place within the four walls of a school setting. Community libraries and churches must be learning centres to help with social distancing. Secondly, employing teachers, there are so many unemployed qualified teachers out there, who can fill the gaps of late, COVID-19 victims. For now, the government is giving schools teacher assistance that have no idea of what is happening in the education sector, leaving many teacher graduates who can be helpful when it comes to digital teaching and learning.

Nomkhu:

COVID-19 is here to stay; technology has to become the order of the day from now onwards. Changing our attitude and learn to live with it will benefit us. We must focus our attention on the changes and be motivated to do better. PLC must advocate for teacher support in the best way possible. Our core mandate as PLC is teaching therefore we must use all the avenues set before us to carry our task to the core. We must be disciplined in order to be effective as PLCs.

Gosakazi:

With the pandemic at hand, things are getting harder and harder. Fourth, fifth to whatever predicted wave, as for me I'll cross that bridge when I get to it. I have accepted that GOVID-19 is here to say. I must learn to live it.

Novangeli:

Teaching and learning does not involve teachers leaders and learners only, it includes parents as well. Whenever the department is developing teachers and learners technologically, parents must be developed as well. They are the ones that will encourage and help learners with their homework and buy them data, if that is possible, because digital networks need data to function.

Vesi decried the employment of the unemployed teachers who have technological skill and the usage of churches and libraries as teaching spaces to minimise physical contacts in schools. Nomkhu had a feeling that COVID-19 will be with us for some time and that teachers need to adapt to the order of the day for teaching and learning to be progressive. Gosakazi did not want to predict anything about the future she will see it when she gets to it. Novangeli's concern was the parental involvement in every skill that the government will do to equip learners and teachers because they are the ones that will help the learners with school work when learners are at home.

Jones et al. (2013) propose the issue of having a procedure to manage time limitations for teacher collaboration learning figuring out how it will happen. These schools further suggests that teacher leaders, and the PLC together with the SMT should arrange time during the working hours for teacher collaborative learning to be viable (Jones et al., 2013). Little (2003) recommends that teacher leadership creates procedures and engagements for restructuring skills for leadership and management in schools. The developmental model on changes in teacher leadership arose as a significant component of the reform procedure and policy in the modern era and the role of teachers as leaders of teaching (Brooks, 2017; Brooks & Mutohar, 2018; Ezzani & Brooks, 2019).

Due to the pandemic online classes are becoming common at the school level. Online courses are designed to ensure the application of online teaching, principles for teachers role, getting to apparatuses to give better input, leveraging of advantages, supporting reasonable teacher-learner ratio, and learners roles in online courses (McPherson & Nunes, 2006: Means et al., 2014). Furthermore teacher leaders must be able to train their PLC members with necessary technological skills, teaching and learning will never be the same again. Moreover, teachers' involvement in PLCs' advance their skills in teaching (Kruse & Johnson, 2017). Through PLC, teachers are accommodating to new concepts and embrace diversity in teaching practices (Lee et al., 2011) to assist in learners' diverse requirements, that assists with propelling learners in learning (Stoll et al., 2006).

4.9 Chapter Summary

This chapter provided a detailed presentation and discussion of the data that was generated from semi-structured interviews, using 8 themes. Data were generated from two peri-urban schools. It was found that, participants were stressed and confused about the impact of COVID-19 in PLCs in the schooling. Because of their socio-economic status in the area technology is not at all helping them with teaching and learning. Their instructional strategies to leverage challenges were not helping, not because of insufficient working material only but because of teacher's attitude as well. Teacher leaders are not supported by PLC members, PLC members have opted for isolation and there is no collaborative teaching happening. The following chapter provides the summary, conclusions and recommendations of the study.

CHAPTER FIVE

SUMMARY, CONCLUSION AND RECOMMENDATIONS

5.1 Introduction

In the preceding chapter, I presented findings that emanated from data collected by means of semi-structured interviews. This chapter presents some lessons learnt from the research process by way of drawing conclusions and making recommendations. I begin the presentation of this conclusions chapter by providing summary of the whole study. However, before I give that summary, I restate the three research questions. This is to remind the readers about what guided this study, and, then, to use these questions to organise the discussion of the conclusions. After discussing the conclusions, I make recommendations directed at various stakeholders.

The research questions of the study:

- What are the teacher leaders' experiences of their role in developing and leveraging professional learning communities in their schools during and post the COVID-19 pandemic?
- What challenges did teacher leaders encounter in developing and leveraging the Professional learning Communities in their schools during and post the COVID-19 pandemic?
- What appropriate strategies did teacher leaders use to mitigate the challenges they encountered when developing and leveraging the Professional Learning Communities in their schools during and post COVID-19 pandemic?

5.2 The study summary

This study is aimed at exploring the role of teacher leaders in developing and leveraging professional learning communities during and post the COVID-19 pandemic. The development and leveraging of professional learning communities (PLCs) and the role that can be played by teacher leaders to develop digital network strategies during and beyond COVID-19 crisis in

Pinetown District was explored. The first chapter is the overview of the problem underpinning the study. The chapter provides the background and the rational for the study. The three research questions that guided the study are also stated. The significance of the study is also stated in terms of which, the importance of understanding the role that can be played by teacher leaders to develop digital network strategies is outlined. The elucidation of key concepts, the outline of the study, the statement of the aims and objectives of the study and demarcation of the study, as well as the demarcation of the study follow. In Chapter One, the unprecedented closure of schools in South Africa due to COVID-19. And its ramifications were explained. The notion of teacher leadership and its possible contribution in the creation and strengthening of the PLCs in terms of responding to the COVID-19 pandemic was introduced in this chapter, and more discussion is needed. Such a detailed discussion is provided in the next chapter, namely, Chapter Two.

The next logical step to take in the research process is the detailed engagement with literature. In this study, literature review and the discussion of the theoretical framework is discussed in Chapter Two. The discussion of Chapter Two begins with the outline of key concepts that underpin the study. This is done by drawing from various scholars and also explaining how these concepts are applied in this study. The next step is about explaining the methodology that was used in generating data that would help address the research questions. This is done in Chapter Three. As part of the descriptions of the methodological issues, this chapter provides a detailed discussion about a variety of methodological issues, and these include the discussion of the research design of the study, an overview of the empirical study, the approach used, the sampling procedures, the allocation of participants to the study, the data generation techniques, as well as the analysis, the trustworthiness issues and ethical considerations. The fourth chapter presents a discussion about the analysis and presentation of the generated data. Verbatim quotes from the research participants are used to support the claims that I make drawing from the analysis of the interviews. The fifth and final chapter presents the conclusions that I draw from the findings presented and discussed in Chapter Four. Recommendations drawn from the conclusions are made.

5.3 Presentation of conclusion

The research findings highlighted that teacher leaders in the participating schools are faced with numerous barriers in promoting collaborative learning through professional Learning

communities. What has also emerged is that, although though the PLC existed in all four schools in one form or the other some teachers still did not want to cooperate with teacher leaders who facilitated the operations of these PLCs. This was more pronounced after the outbreak of COVID-19 pandemic. Operational protocols that were introduced in response to the pandemic's outbreak undermined all the principles of collaborative learning environment. Conclusions are presented and discussed under the following heading; (a) Participants' understanding of professional learning communities (PLC) and their role in developing PLC at the school; (b) The impact of COVID-19 on the functioning of PLCs and promoting cohesion and connecting opportunities during this pandemic; (c) The challenges encountered in developing PLC and 'deprivatise practice' and prevent teacher isolation with the COVID-19 protocol in place—during this time, and (d) The strategies to leverage and maintaining shared routines and the way forward for PLCs post the pandemic.

5.3.1 Participants understanding of professional learning communities (PLC) and their role in developing them at the school

The study is about PLCs and how, through this structure, collaborative learning can proceed during and beyond the pandemic (COVID-19). According to Whitford and Smith (2010), PLCs are a "small groups of educators meeting regularly to engage in systematic peer critique and support by sharing their own professional practices as well as artefacts of student learning". A PLC can be operationalized as educators working collaboratively and interdependently through on-going collective inquiry and action research with an intense focus on student learning (Mattos, DuFour, DuFour, Eaker & Many, 2016). At the centre, of the concept a PLC lays the reason of improving and developing the learners' learning by developing and enhancing teaching practice. One of the assumptions I had before conducting this study was that all teacher leaders had some ideas about what PLCs are. However, one participant had no idea about PLC and declared that she had never heard of the term PLC. However, during our conversation with this participant, it emerged that, in fact, she had some ideas about what constitutes PLCs, although she had not been exposed to this concept. One participant expressed a clear understanding of what a PLC is. The other two participants did not know the term they gave it a tried since they said the term is self-explanatory. When the participants were asked whether they understood their roles as teacher leaders it became evident that they had good understandings of their roles in the PLCs. When looking at the role of teacher leaders, scholars

(Admiraal et al., 2019; Maloney & Konza, 2011; Printy, 2008) emphasised the role of teacher leaders as creators and facilitators of teacher participation in the PLCs. Notably, even the two participants who had said that the term is self-explanatory, expressed their understanding of their role in the PLCs quite eloquently and in a manner that is consistent with the key principles of the concept.

Scholars define leadership as inclusive of both assigned leaders, as well as emergent leaders, who are influential to the group members and of whom group members are receptive of their influence (Northouse, 2019). As teacher leaders continue leading and developing others, leadership skills can be seen and demonstrated. Huijboom et al. (2021) identify three steering factors to the PLC – leadership, collective autonomy, and the facilitation of group dynamic processes. Based on the findings presented in Chapter Four, I can conclude here that the participants as teacher leaders met some of these steering factors. For instance, they did exercise leadership in ensuring that effective teaching and learning occurred before and during the outbreak of COVID-19 pandemic. It is also clear to me that, the outbreak of COVID-19 brought about frustration among them such that the basic notion of PLCs seemed to have eluded them altogether. In short, the challenges were so frustrating that they found it difficult to think through and beyond the pandemic. In terms of facilitating group dynamics processes, they seemed to have made minimum success, especially after the outbreak of the COVID-19 pandemic.

5.3.2 The impact of COVID-19 on the functioning of Professional Learning Communities and promoting coherency and connecting opportunities during this pandemic

One can tell that COVID-19 has impacted negatively in these schools because the data indicates that the four teacher leaders in both schools share the similar sentiment about the functioning of PLCs in their schools. They mentioned that members have bailed out from the PLC and they said that members are demoralised. That has led to the downfall of communication among PLC members and they stressed that their PLC were inefficient because their main focus was on learners not on the PLC. There were instances where teachers had positive attitudes towards the PLC but they were no longer not willing to collaborate if they feel there them had restricted and limited autonomy. These issues are also echoed by other scholars (Al Shammari et al. 2020; Thornton & Cherrington (2014). These scholars emphasise the point that the role of teacher

leaders is fundamental for moulding trust relations and restrictions undermined their independence and trust amongst the members of the PLCs.

The participants' reported that their instructional activities as PLC had become ineffective and inefficient due to low level of cooperation among PLC members. Teacher leaders working with teachers on classroom observations and constructive feedback conversations were put on hold because of the pandemic that was reportedly escalating. Cohesion and staff morale plummeted. The worldwide perspective on online education has been seen as a good-to-have alternative but in all honesty it was not a model to ensure consistence of instructional activities (Ribeiro, 2020). Other schools find in easy to transform from traditional ways of teaching whilst others the migration process led them to a crisis due to the pandemic, as cited by Hodges et al. (2020) and Manfuso (2020.

Literature indicates that teacher leaders are responsible to encourage and motivate teachers to work together and working hard by sharing information. The top-down traditional ways of professional development are often occasional and unclear at times to impact teachers. PLCs provide a better option for teacher development which is more peer centred. Even though it does not automatically create more cohesion and connected opportunities but it is theirs and they are able to develop it in the way it suits best. The pride in PLCs is in ownership. One can claim that there is no coherency or connection in the two schools. Teachers have taken a 'backseat' when it comes to working with others. The findings have indicated that those who contracted the virus were their coherency collapsed due to stigmatised instead of exercising compassion and understanding. What I can conclude on this matter is that the participants seemed to lose a lot regarding the existence of the PLCs and how they can be strengthened. It is evident that cohesion which should have been developed and enhanced was lost completely

5.3.3. The Challenges encountered in developing the Professional Learning Communities and 'deprivatise practice' and prevent teacher isolation with the COVID-19 protocol in place

Teacher leaders from the two schools saw individualism as a challenge that leads to dysfunctionality of the PLC in their schools and their moral had declined because of the level of commitment for PLC members. The participants expressed unhappiness about the performance level of the PLC in their respective schools. Teachers leaders in this report felt

that if the government has technologically equip teachers before the outbreak of COVID-19, major challenges would have been avoided. Alhouti, (2020) argues that when the Department of Education fails to provide financial help for public schools, the implementation of online learning fails as well. Existing studies (Maloney & Konza, 2011); Printy (2008); Ratts et al., 2015; Thompson et al., 2004; Watson, (2014) provided evidence that sharing expertise and knowledge fosters growth and learning and solve teaching issues for teachers in PLCs and it also improve learners' learning and school performance.

The introduction of COVID-19 protocols was meant for good outcomes but because people have different views it ended up being the one that propelled individualism and isolation in these two schools. The participants aired their frustration on COVID-19 protocols that brought high level of isolation. Teachers who have no experience in online teaching or have difficulty in operating technology and information gadgets usually have problems and challenges in smoothness of instructional activities What I can conclude from the findings is that the participants were so absorbed by the challenges they encountered to the point that teacher leaders seemed to have forgotten that PLCs stand a better chance of mitigating the challenges posed by the pandemic. The bigger picture was completely lost. For instance, online learning was meant to ensure that learning can continue without people having to meet physically. However, the participants did not see that point, and preferred physical presence, and without it, life seemed to have come to an end for them. The second point is that the fact that due to the pandemic, a number of teachers have become accustomed to having the privacy and independence to teach in the manner in which they want, they did do not want others' contribution and interference in their teaching. The conclusion I make is that the participants did not understand the strength of the PLC in terms of ensuring that connectivity can continue even though people are working in their own individual spaces. Again, I can conclude that the participants' understanding of the PLCs was limited, especially when it comes to what the PLCs can contribute. The threats to the quality teaching and learning environment as posed by COVID-19 pandemic can arguably be addressed through the use of the PLCs. Therefore, the collapse of the work of the PLCs suggests that teacher leaders may not have clearly understood the essence and the benefits of the PLCs.

5.3.4 The strategies to leverage and maintaining shared routines and the way forward for PLCs post the pandemic.

The major part of teacher support is professional development. All four participants concurred that professional development of the teaching staff was an essential element in uplifting the standard of teaching and learning. The most commonly cited by almost all four of them was the use of workshops whereby different experts can be invited to the school to share skills that most staff members or even for one teacher's needs. The data indicates that the participants concurred that traditional or face-to-face ways of holding meetings are slowly expiring. It is clear that they all see technology is one of the strategies that can be used to deal with the barriers that they had encountered to leverage PLC especially during these trying times. They have the mentality that no matter how few they can be but they were willing to push forward in terms of maximising their efforts to strengthen the operation of the PLCs. However, the data revealed that some participants did not cooperate with others despite obvious benefits to do so. The irony is that, the participants seemed not to recognise that the PLCs stood a better chance to address most of the challenges posed by both the pandemic in terms of isolation and online, remote teaching, as well as the challenges relating to infrastructure at school and the teachers' skills deficit. The benefits of digital technology and its integration in teaching and learning has been acknowledged by scholars (Ala-Mutka et al., 2008; Bozkurt & Sharma, 2020; Kaden, 2020; Omotayo & Haliru, 2020; Ribeiro, 2020).

Drawing from the findings and discussions as presented in Chapter Four, I can conclude that teacher leaders in this study did not have a clear and effective strategy to leverage and maintain shared routines and the way forward for PLCs post the pandemic. The data has shown that the PLCs were collapsing and they did not put up any efforts or resistance to this other than the laments and frustrations about what was happening. With regards to the way forward for the PLCs, it is evident that they had no clear plan other than wishful thinking and hope that things will be better. My view is that what the teacher leaders were proposing as a way forward did not show that they had a clear understanding of the PLCs, and to work with the teaching staff to build confidence in this important structure, and what they stood to achieve if the PLCs were strengthened.

5.4 Recommendations

Based on the findings and conclusions made, I have a number of recommendations which I believe will benefit knowledge area, including stakeholders involved in facilitating effective teaching and learning in schools. The first recommendation is directed at teacher leaders as

facilitators of continuous professional development. It is therefore, recommended that teacher leaders should strengthen their efforts in ensuring that teachers work together as a community of professional who continuously hone their teaching skills despite various challenges in the environment inside and outside the school premises.

Secondly, the findings have demonstrated that issues of technology and its integration to curriculum delivery are a current reality and a feature for the future. It has also emerged that many teachers have skills deficit when it comes to the use of computers and online teaching and learning scenarios. Directly linked to that is a fact that many schools from deprived school contexts do not have requisite infrastructure as alluded to by various scholars. Therefore, the second recommendation teacher leaders need to maximise their efforts in supporting teachers to embrace the use of technology to enhance teaching and learning. They need to do more work in terms of sensitising their staff in terms of honing their skills in this regard and also put more efforts towards exposing them to training in these skills.

The third finding demonstrated that isolation due to COVID-19 pandemic has contributed to the decline in teacher participation in the PLCs. The teachers did not realise that PLCs are actually, vehicles that can enhance improvement in teaching and learning even in hostile conditions such as those imposed by COVID-19. In fact, the lack of interaction between teachers and learners because of the effect of COVID-19 pandemic, which appeared to be the main challenge, should not be seen as impediment to the sustainability of the PLCs. Such realities need to raise the consciousness of teacher leaders about the need to brainstorm ideas that can help to adapt to new ways of handling classes, mobilising resources and exposing staff to training that will ensure accomplishments of school goals.

5.5 Chapter Summary

Chapter Five has presented the results of the study based on the findings that were discussed in the previous chapter. The chapter began with a summary before conclusions reached were discussed. Recommendations made, based on the conclusions have been made.

References

- Aarons, G. A., Ehrhart, M. G., Farahnak, L. R., & Hurlburt, M. S. (2015). Leadership and organizational change for implementation (LOCI): a randomized mixed method pilot study of a leadership and organization development intervention for evidence-based practice implementation. *Implementation Science*, 10(1), 1-12.
- Abbott, A. (2014). The system of professions: An essay on the division of expert labour. University of Chicago press.
- Abdulamir, A. S., & Hafidh, R. R. (2020). The Possible Immunological Pathways for the Variable Immunopathogenesis of COVID—19 Infections among Healthy Adults, Elderly and Children. *Electron J Gen Med.* 17 (4): 2020.
- Abu-Bakar, S. H., Jamian, J. J. Muhtazaruddin, M. N., Bani, N. A., Muhammad-Sukki, F., & Mas' ud, A. A. (2016). Impact of distributed generation modes in optimal coordination between dG and capacitor simultaneously.
- Adarkwah, M. A. (2021). "I'm not against online teaching, but what about us?": ICT in Ghana post Covid-19. *Education and Information Technologies*, 26(2), 1665-1685
- Admiraal, W Wang, J., &.Tigelaar, D. E., (2019). Connecting rural schools to quality education: Rural teachers' use of digital educational resources. *Computers in Human Behaviour*, 101, 68-76.
- Aksal, F. A., & Gazi, Z. A. (2015). Examination on ICT integration into special education schools for developing countries. *Turkish Online Journal of Educational Technology*-TOJET, 14(3), 70-72.
- Ala-mutka, K.; Punie, Y. & Redecker, C. (2008). *Digital competence for lifelong learning*. Luxembourg: Office for Official Publications of the European Communities
- Alhouti, I. (2020). Education during the pandemic: the case of Kuwait. *Journal of Professional Capital and Community*.
- Al-Shammari, A. A., Alahmad, B., Bennakhi, A., Al-Mulla, F., & Ali, H. (2020). Fasting blood glucose and COVID-19 severity: nonlinearity matters. *Diabetes Care*, 43(12), 3113-3116.
- AM Vermeulen, P., Zietsma, C., Greenwood, R., & Langley, A. (2016). Strategic responses to institutional complexity. *Strategic Organization*, *14*(4), 277-286.

- Andriani, S., Kesumawati, N., & Kristiawan, M. (2018). The influence of the transformational leadership and work motivation on teacher's performance. *International Journal of Scientific & Technology Research*, 7(7), 19-29.
- Bakshi, S., & Awasthi, K. (2019). Leading Schools Moving from Issues and Challenges. *EDUCATION*, 45(3).
- Baldwin, C. K., & Wilder, Q. (2014). Inside quality: Examination of quality improvement processes in afterschool youth programs. *Child & Youth Services*, 35(2), 152-168.
- Banasik, M. D., & Dean, J. L. (2016). Non-tenure track faculty and learning communities: Bridging the divide to enhance teaching quality. *Innovative Higher Education*, 41(4), 333-342.
- Baran, E. (2016). Examining the benefits of a faculty technology mentoring program on graduate students' professional development. *Journal of Digital Learning in Teacher Education*, 32(3), 95-104.
- Bardesi, H., Al-Mashaikhi, A., Basahel, A., & Yamin, M. (2021). COVID-19 compliant and cost effective teaching model for King Abdulaziz University. *International Journal of Information Technology*, 13(4), 1343-1356.
- Basilaia, G., & Kvavadze, D. (2020). Transition to online education in schools during a SARS-CoV-2 coronavirus (COVID-19) pandemic in Georgia. *Pedagogical Research*, 5(4).
- Battiste, M., Chomsky, N., Denzin, N. K., Fine, M., Gill, R., Grande, S., ... & Weistheimer, J. (2018). *Dissident knowledge in higher education*. University of Regina Press.
- Beaunoyer, E.; Dupéré, S.; Guitton, M.J. COVID-19 and Digital Inequalities: *Reciprocal Impacts and Mitigation Strategies*. Comput. Hum. Behav. 2020, 111, 106424.
- Beddoes, Z. E., Prusak, K. A., & Barney, D. (2019). Taking the helm: Physical educators managing change through professional learning communities. Quest, 71(4), 479-496.
- Bener, E., & Yildiz, S. (2019). The use of blog activities to promote reflection in an ELT practicum. *Australian Journal of Teacher Education (Online)*, *44*(8), 38-56.
- Bennett, B., & Bromen, K. (2019). Growing Teacher Leaders from the Ground Up.

 International Journal of Teacher Leadership, 10(2), n2
- Bennett, S., Maton, K., & Kervin, L. (2008). The 'digital natives' debate: A critical review of the evidence. *British journal of educational technology*, 39(5), 775-786.
- Berg, B. L. (2014). Howard Lune.(2014). *Qualitative Research Methods for the Social Sciences*. Boston: Allyn and Bacon.

- Berg, J. H., Carver, C. L., & Mangin, M. M. (2014). Teacher leader model standards: Implications for preparation, policy, and practice. *Journal of Research on Leadership Education*, 9(2), 195-217.
- Bertram, C., & Christiansen, I. (2014). *Understanding research*. *An introduction to reading research*. Pretoria: Van Schaik Publishers.
- Bhengu, T.T. (2021). Can Decentralisation of Structures and Devolution of Governance power enable an effective response to COVID-19 Disruptions? Lessons from the South African Education system. *Alternations*, 28(1), 106-136.
- Bhengu, T.T., & Blose, S.B. (2022). Developing a collaborative teacher learning culture: Perspectives of three secondary school principals in the Pinetown District. *Journal of Educational Studies*, 21(1),51-66.
- Black, S., Spreen, C.A., & Vally, S. (2020). Education, COVID-19 and Care: Social Inequality and Social Relations of Value in South Africa and the United States. *Southern African Review of Education*, 26(1), 40-61.
- Blömeke, S., & Delaney, S. (2014). Assessment of teacher knowledge across countries: A review of the state of research. *International Perspectives on Teacher Knowledge, Beliefs and Opportunities to Learn*, 541-585.
- Blose, S., & Khuzwayo, N. Q. (2020). Teacher leadership in action: An inquiry into the lived experiences of subject heads in secondary schools. *International Journal of Leadership in Education*, 1-16.
- Bobley, L. & Best, R. (2021). Teacher Preparation during COVID-19: A Shift from Face-to-Face to Remote Field Experiences and Student Teaching. INTED2021, *15th Annual International Technology, Education and Development Conference*. IATED Academy, Spain.
- Bolden, 2011; Bolden, R. (2011). Distributed leadership in organizations: A review of theory and research. *International Journal of Management Reviews*, 13(3), 251-269.
- Bozkurt, A., & Sharma, R. C. (2020). Education in normal, new normal, and next normal: Observations from the past, insights from the present and projections for the future. *Asian Journal of Distance Education*, 15(2), i-x
- Breitkopf, M. (2018). But this is how we have always done it: Overcoming resistance to change in the quest to support distance learners. *Journal of Library & Information Services in Distance Learning*, 12(3-4), 148-164.
- Brooks, D. C., & Pomerantz, J. (2017). ECAR study of undergraduate students and information technology. *EDUCAUSE Centre for Analysis and Research*.

- Brooks, M. C., & Mutohar, A. (2018). Islamic school leadership: A conceptual framework. Journal of Educational Administration and History, 50(2), 54-68.
- Brown, B. B. (2004). Adolescents' relationships with peers.
- Brown, C., & Poortman, C. L. (Eds.). (2018). Networks for learning: *Effective collaboration* for teacher, school and system improvement. Routledge.
- Brown, S., (2020). For both new and existing staff in HE, this book provides a handbook on learning to teach. *Learning, teaching and assessment in higher education:* global perspectives
- Brynard, D. J., Hanekom, S. X., & Brynard, P. (2014). *Introduction to research*. Pretoria: Van Schaik.
- Busch, J. R., O'Brien, T. P., & Spangler, W. D. (2005). Increasing the quantity and quality of school leadership candidates through formation experiences. *Journal of Leadership & Organizational Studies*, 11(3), 95-108.
- Bush, T. (2015). Teacher leadership: Construct and practice. *Educational Management Administration & Leadership*, 43(5), 671-672.
- Bush, T., & Glover, D. (2016). School leadership and management in South Africa: Findings from a systematic literature review. *International journal of educational management*.
- Butcher, J. (2020). Public-private virtual-school partnerships and federal flexibility for schools during COVID-19. *Special Edition Policy Brief*.
- Cambum, E., Rowan, B., & Taylor, J. E. (2003). Distributed leadership in schools: The case of elementary schools adopting comprehensive school reform models. *Educational Evaluation and Policy Analysis*, 25(A), 347-373.
- Cansoy, R., & Parlar, H. (2018). Examining the relationships between the level of schools for being professional learning communities and teacher professionalism. *Malaysian Online Journal of Educational Sciences*, 5(3), 13–27.
- Capra, F.(1996)Paperback. Book Condition: new. fondofomentostamarta.utbvirtual.education, The Web of Life: A New Synthesis of Mind and Matter, *HarperCollins Publishers Fritj* of Capra,
- Chalk C. M., Pastor, J., Peakall, J., Borman, D. J., Sleigh, P. A. Murphy, W. & Fuentess, R., (2020). Stress-Particles Smoothed Particle Hydrodynamics; An application to the failure and post-failure behaviour of slopes. *Computer Methods in Applied Mechanics* and Engineering, 336,113034

- Chatturgoon, S. (2008). An exploration into teachers' perceptions on teacher leadership and their motivational levels to engage in leadership roles at school (Doctoral dissertation).
- Cheng, X., Pellegrini, M., Zhou, L., & Cheung, A. C. (2020). Not Only Survival but Stronger: The Impact of Alarming Invader of SARS-CoV-2 on Global Education. *Science Insights Education Frontiers*, 7(2), 835-860.
- Cheong, M., Spain, S. M., Yammarino, F. J., & Yun, S. (2016). Two faces of empowering leadership: Enabling and burdening. *The Leadership Quarterly*, 27(4), 602-616.
- Chigona, A., Chigona, W., & Davids, Z. (2010, March). Motivating factors: Educators' use of ICT in schools in disadvantaged areas in the Western Cape. In *International Conference on Information Management and Evaluation* (p. 74). Academic Conferences International Limited.
- Chigona, W., Lekwane, O., Westcott, K., & Chigona, A. (2011). Uses, benefits and challenges of public access points in the face of growth of mobile technology. *The Electronic Journal of Information Systems in Developing Countries*, 49(1), 1-14.
- Chiniara, M., & Bentein, K. (2016). Linking servant leadership to individual performance: Differentiating the mediating role of autonomy, competence and relatedness need satisfaction. *The Leadership Quarterly*, 27(1), 124-141.
- Chothia, A. (2020). Coronavirus outbreak closes all South African universities, colleges. *The South African*.
- Christiansen A., Prescott T., & Ball J., (2014). Learning in action: Developing safety improvement capabilities through action learning. *Nurse Education Today*, 34(2), 243-247.
- Ciulla, J. B. (2009). Ethics and leadership effectiveness. *Discovering leadership*, 325-340.
- Clarke, V., & Braun, V. (2013). Teaching thematic analysis: Overcoming challenges and developing strategies for effective learning. *The psychologist*, 26(2).
- Cober, R., Tan, E., Slotta, J., So, H-J., & Könings, K.D. (2015). Teachers as participatory designers: two case studies with technology-enhanced learning environments Springer *Science+Business Media Dordrecht* Instr Science, 3,203–228.
- Cochran-Smith, M., & Lytle, S. L. (2009). Teacher research as stance. *The Sage Handbook of Educational Action Research. London: Sage*, 39-49.
- Cochran-Smith, M., & Lytle, S. L. (Eds.). (1993). *Inside/outside: Teacher research and knowledge*. Teachers College Press.
- Cohen, E., Hoz, R., & Kaplan, H. (2013). The practicum in preservice teacher education: a review of empirical studies. *Teaching Education*, 24(4), 345-380.

- Cohen, L., Manion, L. & Morrison, K. (2011). *Research methods in education* (7th Ed.). New York: Routledge.
- Cohen, L., Manion, L., & Morrison, K. (2018). *Research methods in education* (8th Ed.), London: Routledge, United Kingdom.
- Cole, M. (2006). Qualitative research: a challenging paradigm for infection control. *British Journal of Infection Control*, 7(6), 25-29.
- Coman, C.; Tîru, L.G.; Meses, an-Schmitz, L.; Stanciu, C.; Bularca, M.C. (2020), Online Teaching and Learning in Higher Education during the Coronavirus Pandemic: Students' Perspective. *Sustainability*, 12, 10367.
- Conrad, R. M., & Donaldson, J. A. (2011). Engaging the online learner: Activities and resources for creative instruction (Vol. 38). John Wiley & Sons
- Cook, L. H., Boe, E. E., & Sunderland, R. J. (2008). Teacher turnover: Examining exit attrition, teaching area transfer, and school migration. *Exceptional Children*, 75(1), 7-31.
- Cooper, A., MacGregor, S., Shewchuk, S. (2020). A research model to study research-practice partnerships in education. *Journal of Professional Capital and Community*
- Couto, K. C., Moura Lorenzo, F., Tagliabue, M., Henriques, M. B., & Freitas Lemos, R. (2020). Underlying principles of a Covid-19 behavioral vaccine for a sustainable cultural change. *International Journal of Environmental Research and Public Health*, 17(23), 9066.
- Creswell, J. W. (2013). *Qualitative Inquiry and Research design: Choosing among Five Approaches*. Thousand Oaks, CA: SAGE Publications.
- Creswell, J. W. (2014). *Research design: Qualitative, Quantitative and Mixed Method Approaches* (4th Ed). London: SAGE Publications.
- Creswell, J.W. (2009). *Research Design: Qualitative, Quantitative, and Mixed Methods Approaches*, 3rd edition. Thousand Oaks, CA: SAGE Publications.
- Curry, M. (2008). Critical friends groups: The possibilities and limitations embedded in teacher professional communities aimed at instructional improvement and school reform. *Teachers college record*, 110(4), 733-774.
- Danielson, C. (2006). Teacher leadership that strengthens professional practice. ASCD.
- Darling-Hammond & Hyler, (2020) L. Darling-Hammond, K. Hammerness, P. Grossman, F. Rust and L. Shulman, "Chapter 11 The design of teacher education programs" in Preparing Teachers for a changing world: *What teachers should be learning and be able*

- *to* (L. Darling-Hammond and J. Bransford, eds.), pp. 391-441, San Francisco: Jossey Bass, 2005.
- De Villiers, E., & Pretorius, S. G. (2011). Democracy in schools: are educators ready for teacher leadership?. *South African Journal of Education*, *31*(4), 574-589.
- DeMatthews, D. E. (2014). Principal and teacher collaboration: An exploration of distributed leadership in professional learning communities. *International Journal of Educational Leadership and Management*, 176-206
- DeMore Palmer, C. F. (2011). *Insiders' voices: A Phenomenological Study Of Informal Teacher Leadership From The Perspective Of Those Who Choose To Lead* (Doctoral dissertation, University of Pittsburgh).
- Denzin, N. K. (2000). Aesthetics and the practices of qualitative inquiry. *Qualitative Inquiry*, 6(2), 256-265.
- Denzin, N.K., & Lincoln, Y. S. (2000). *Handbook of qualitative research*, (2nd ed.). Thousand Oaks, CA: SAGE Publishers.
- DeWitt, P. M. (2016). Collaborative leadership: Six influences that matter most. Corwin Press.
- Dinc, E. (2019). Prospective teachers' perceptions of barriers to technology integration in education. *Contemporary Educational Technology*, 10(4), 381–398.
- Djalante, R., Lassa, J., Setiamarga, D., & Rafiana, I. (2020). Review and analysis of current responses to COVID-19 in Indonesia: Period of January to March 2020.

 https://www.ingsa.org/covidtag/academic-papers/djalante-indonesia/. Accessed 15 May 2020.
- Donaldson, M. L., Johnson, S. M., Kirkpatrick, C. L., Marinell, W. H., Steele, J. L., & Szczesiul, S. A. (2008). Angling for access, bartering for change: How second-stage teachers experience differentiated roles in schools. *Teachers College Record*, 110(5), 1088–1114.
- Dreesen, T., Akseer, S., Brossard, M., Dewan, P., Giraldo, J. P., Kamei, A., & Ortiz, J. S. (2020). Promising practices for equitable remote learning: Emerging lessons from COVID-19 education responses in 127 countries
- Du Toit, A. (2020). Outbreak of a novel coronavirus. *Nature Reviews Microbiology*, 18(3), 123-123.
- DuFour R., DuFour R., Eaker R., & Many, T. (2006). Learning by doing: *A handbook for professional learning communities at work*. Bloomington, IN: Solution Tree.

- DuFour, R. (2007). Professional learning communities: A bandwagon, an idea worth considering, or our best hope for high levels of learning? *Middle School Journal*, 39(1), 4-8.
- DuFour, R., & Marzano, R. J. (2011). *Leaders of learning: How district, school, and classroom leaders improve student achievement.* Solution Tree Press.
- Elmore, R. F. (2002). Bridging the gap between standards and achievement: The imperative for professional development in education. *Secondary lenses on learning participant book: Team leadership for mathematics in middle and high schools*, 313-344.
- Ertmer, P. A. (1999). Addressing first-and second-order barriers to change: Strategies for technology integration. *Educational Technology Research and Development*, 47(4), 47-61.
- Evans, S. M., Gill, M. E., & Marchant, J. (1996). Schoolchildren as educators: the indirect influence of environmental education in schools on parents' attitudes towards the environment. *Journal of Biological Education*, *30*(4), 243-248.
- Ezzani, M., & Brooks, M. (2019). Culturally relevant leadership: Advancing critical consciousness in American Muslim students. *Educational Administration Ouarterly*, 55(5), 781-811.
- Fairman, J. C., & Mackenzie, S. V. (2012). Spheres of teacher leadership action for learning. *Professional Development in Education*, *38*(2), 229-246.
- Feldman, J. (2020). The role of professional learning communities to support teacher development: A social practice theory perspective. *South African Journal of Education*, 40(1), 1-8.
- Fernandez, A. A., & Shaw, G. P. (2020). Academic leadership in a time of crisis: The Coronavirus and COVID-19. *Journal of Leadership Studies*, *14*(1), 39-45.
- Fletcher, J., & Nicholas, K. (2016). What can school principals do to support students and their learning during and after natural disasters? *Educational Review*, 68(3), 358-374.
- Flick, U. (2018). Designing qualitative research. SAGE.
- Fullan, M. (2001). The new meaning of educational change. Routledge.
- Fullan, M., & Hargreaves, A. (2016). *Call to action: Bringing the profession back in*. Oxford, OH: Learning Forward.
- Gabriel, A. S., Daniels, M. A., Diefendorff, J. M., & Greguras, G. J. (2015). Emotional labor actors: a latent profile analysis of emotional labor strategies. *Journal of applied psychology*, 100(3), 863.

- Garet, M. S., Porter, A. C., Desimone, L., Birman, B. F., & Yoon, K. S. (2001). What makes professional development effective? Results from a national sample of teachers. *American Educational Research Journal*, 38(4), 915-945.
- Ghemawat, P. (2003). Semi globalization and international business strategy. *Journal of International Business Studies*, 34(2), 138-152.
- Gilligan, C. (2011). Ethics of care. Entrevista concedida a Webteam em, 21.
- Glisson, C., & Schoenwald, S. K. (2005). The ARC organizational and community intervention strategy for implementing evidence-based children's mental health treatments. *Mental health services research*, 7(4), 243-259.
- Gosselin, K. P., Northcote, M., Reynaud, D., Kilgour, P., Anderson, M., & Boddey, C. (2016). Development of an Evidence-Based Professional Learning Program Informed by Online Teachers' Self-Efficacy and Threshold Concepts. *Online Learning*, 20(3), 178-194.
- Grant, A. M. (2008). Does intrinsic motivation fuel the prosocial fire? Motivational synergy in predicting persistence, performance, and productivity. *Journal of applied psychology*, 93(1), 48.
- Grant, A. M. (2012). Leading with meaning: Beneficiary contact, prosocial impact, and the performance effects of transformational leadership. *Academy of Management Journal*, 55(2), 458-476.
- Grant, C. (2006). Emerging voices on teacher leadership: Some South African views. *Educational Management Administration & Leadership*, *34*(4), 511-532.
- Grant, C. A. (2008). Teacher capacity. In M. Cochran- Smith, S. Feiman-Nemser D. J. McIntyre (Eds.), *Handbook of Research on Teacher Education, Enduring* Questions in Changing Contexts (pp. 127-133). New York: Routledge/Tayler Francis Group
- Grant, C., & Singh, H. (2009). Passing the buck: this is not teacher leadership! *Perspectives in Education*, 27(3), 289-301.
- Grant, M. M. (2019). Difficulties in defining mobile learning: Analysis, design characteristics, and implications. *Educational Technology Research and Development*, 67(2), 361-388.
- Grant, S. G., Swan, K., & Lee, J. (2017). *Inquiry-based practice in social studies education:*Understanding the inquiry design model. Routledge.
- Harris, A. (2013). Distributed leadership: Friend or foe? *Educational Management Administration & Leadership*, 41(5), 545-554.

- Harris, A., & Jones, M. (2017). Leading in context: Putting international comparisons into perspective. *School Leadership & Management*, *37*(5), 431-433.
- Harris, A., & Jones, M. (2020). COVID 19–school leadership in disruptive times. *School Leadership & Management*, 40(4), 243-247.
- Heiner, M., Gilbert, D., & Donaldson, R. (2008, June). Petri nets for systems and synthetic biology. In *International school on formal methods for the design of computer, communication and software systems* (pp. 215-264). Springer, Berlin, Heidelberg.
- Hennessy, S., Ruthven, K., & Brindley, S. U. E. (2005). Teacher perspectives on integrating ICT into subject teaching: commitment, constraints, caution, and change. *Journal of curriculum studies*, *37*(2), 155-192.
- Henning, E., van Rensburg, W., & Smit, B. (2004). *Finding your way in qualitative research*. Pretoria: Van Schaik.
- Hew, K. F., & Brush, T. (2007). Integrating technology into K-12 teaching and learning: Current knowledge gaps and recommendations for future research. *Educational Technology Research and Development*, 55(3), 223-252.
- Hiebert, J., & Carpenter, T. P. (1992). Learning and teaching with understanding. *Handbook of research on mathematics teaching and learning: A project of the National Council of Teachers of Mathematics*, 65-97.
- Hodges, C. B., & Fowler, D. J. (2020). The COVID-19 Crisis and Faculty Members in Higher Education: From Emergency Remote Teaching to Better Teaching through Reflection. *International Journal of Multidisciplinary Perspectives in Higher Education*, 5(1), 118-122.
- Hodges, C. B., Moore, S., Lockee, B. B., Trust, T., & Bond, M. A. (2020). *The difference between emergency remote teaching and online learning*.
- Holden, H., & Rada, R. (2011). Understanding the influence of perceived usability and technology self-efficacy on teachers' technology acceptance. *Journal of Research on Technology in Education*, 43(4), 343-367.
- Holmén, J., Adawi, T., & Holmberg, J. (2021) Student-Led Sustainability Transformations: *Employing Realist Evaluation to Open the Black Box of Learning in a Challenge Lab Curriculum. Int. J. Sustain. High. Educ.* 2021, 22, 1–24.
- Hubers, M. D., Poortman, C. L., Schildkamp, K., Pieter's, J. M., & Handelzalts, A. (2016).
 Opening the black box: Knowledge creation in data teams. *Journal of professional capital and community*.

- Huijboom, F., Van Meeuwen, P., Rusman, E., & Vermeulen, M. (2021). Keeping track of the development of professional learning communities in schools: the construction of two qualitative classification instruments. *Professional development in education*, 47(4), 667-683.
- Jackson, T., Burrus, J., Bassett, K., & Roberts, R. D. (2010). Teacher leadership: An assessment framework for an emerging area of professional practice. *ETS Research Report Series*, 2010(2), i-41.
- Jin, Y., & Redish, T. (2020). Professional development for remote learning in teacher education to support teacher educators and preservice teachers during the COVID-19 pandemic. Technology and teacher education during the COVID-19 pandemic: Stories from the field, 473-477.
- Jones, B., & Iredale, N. (2010). Enterprise education as pedagogy. *Education+ training*.
- Kaden, U. (2020). COVID-19 school closure-related changes to the professional life of a K–12 teacher. *Education Sciences*, *10*(6), 165.
- Kajee, L. (2011). Literacy journeys: home and family literacy practices in immigrant households and their congruence with schooled literacy. *South African Journal of Education*, *31*(3), 434-446.
- Kali, M. (2020). Challenges and prospects of decentralisation in Lesotho. *International Journal of Humanities, Arts and Social Studies*, 5(2), 11-20.
- Katzenmeyer, M., & Moller, G. (2009). Awakening the sleeping giant: Helping teachers develop as leaders. Corwin Press
- Keppell, M., O'Dwyer, C., Lyon, B., & Childs, M. (2010). Transforming distance education curricula through distributive leadership. *Australasian Journal of Educational Technology*, 26(8).
- Kesson, K. (2020). Three Scenarios for the Future of Education in the Anthropocene. *The Journal of Future Studies*. Retrieved from https://jfsdigital.org/2020/04/12/three-scenarios-for-the-future-of-education-in-the-anthropocene.
- Khumalo, N. (2008). The implementation of Integrated Quality Management System challenges facing the DSGing in Vryheid District of KwaZulu-Natal. *Unpublished mini-dissertation*. *University of Johannesburg*: Johannesburg.
- Ko, S., & Rossen, S. (2017). *Teaching online: A practical guide*. Routledge.
- Koonce, M., Pijanowski, J. C., Bengtson, E., & Lasater, K. (2019). Principal engagement in the professional development process. *NASSP Bulletin*, 103(3), 229-252.

- Korstjens, I., & Moser, A. (2018). Series: Practical guidance to qualitative research. Part 4: Trustworthiness and publishing. *European Journal of General Practice*, 24(1), 120-124
- Kotter, J. (1996). Leading change. Boston, MA: Harvard Business School Pres
- Kruse, S. D., & Johnson, B. L. (2017). Tempering the normative demands of professional learning communities with the organizational realities of life in schools: Exploring the cognitive dilemmas faced by educational leaders. *Educational Management Administration & Leadership*, 45(4), 588-604.
- Kumar, M. (2011). Impact of the evolution of smart phones in education technology and its application in technical and professional studies: Indian perspective. *International Journal of Managing Information Technology (IJMIT) Vol, 3*.
- Kumar, S. (2014). Establishing linkages between emotional intelligence and transformational leadership. *Industrial Psychiatry Journal*, 23(1), 1.
- Kvale, S. (1996). The 1,000-page question. *Qualitative inquiry*, 2(3), 275-284.
- Lave, J., & Wenger, E. (1991). Learning in doing: Social, cognitive, and computational perspectives. *Situated learning: Legitimate peripheral participation*, 10.
- Lavonen, J. (2017). Governance decentralisation in education: Finish Innovation in Education. *Revista de Educacion a Distancia* (RED), 53(1), 1-22.
- Lee, J. C. K., Zhang, Z., & Yin, H. (2011). A multilevel analysis of the impact of a professional learning community, faculty trust in colleagues and collective efficacy on teacher commitment to students. *Teaching and teacher education*, 27(5), 820-830.
- Leithwood, K., & Jantzi, D. (2000). The effects of transformational leadership on organizational conditions and student engagement with school. *Journal of Educational Administration*.
- Leithwood, K., & Riehl, C. (2003). What we know about successful school leadership.
- Leithwood, K., Harris, A., & Hopkins, D. (2008). Seven strong claims about successful school leadership. *School Leadership and Management*, 28(1), 27-42.
- Leithwood, K., Harris, A., & Hopkins, D. (2019). Seven strong claims about successful school leadership revisited. *School Leadership and Management*, 40(4), 1–18. https://doi.org/10.1080/13632434.2019.1 596077
- Leithwood, K., Harris, A., & Hopkins, D. (2020). Seven strong claims about successful school leadership revisited. *School Leadership & Management*, 40(1), 5-22.

- Levine, A. (2015) *The Education Schools Project. Educating school teachers*, Washington, DC.
- Levine, P., & Kawashima-Ginsberg, K. (2015). Civic Education and Deeper Learning.

 Deeper Learning Research Series. *Jobs for the Future*.
- Lincoln, Y. S., & Guba, E. G. (1985). *Naturalistic inquiry*. Newbury Park, CA: SAGE Publications
- Lowrie, M. (2020). Care and security in Vergils Aeneid: an analysis of the politics of empire.

 In *Paradox and Power in Caring Leadership*. Edward Elgar Publishing.
- Lynch, D. (2012). Preparing Teachers in Times of Change. Lulu. com.
- Macgrath, B., Aslam, M., & Johnson, D. (2019). Systems research in education: Designs and methods. *Research in Comparative and International Education*, *14*(1), 7-29.
- Makoelle, T. M., & Makhalemele, T. (2020). Teacher leadership in South African schools. *International Journal of Management in Education*, *14*(3), 293-310.
- Maloney, C., & Konza, D. (2011). A case study of teachers' professional learning: Becoming a community of professional learning or not? *Issues in Educational Research*, 21(1), 75-87.
- Manfuso, L. G. (2020). From emergency remote teaching to rigorous online learning. *Retrieved August*, *10*, 2020.
- Maree, J. G. (2015). Research on life design in (South) Africa: A qualitative analysis. *South African Journal of Psychology*, 45(3), 332-348.
- Marongwe, N., & Garidzirai, R. (2021). Together but not together: Challenges of remote learning for students amid the COVID-19 pandemic in rural South African universities. *Research in Social Sciences and Technology*, 6(3), 213-226.
- Mattos, M. W., DuFour, R., DuFour, R., Eaker, R., & Many, T. W. (2016). *Learning by doing: a handbook for professional learning communities at work*. Solution Tree Australia Pty Limited.
- McPherson, M., & Nunes, M. B. (2006). Organisational issues for e-learning: Critical success factors as identified by HE practitioners. *International Journal of Educational Management*.
- Means, B., Bakia, M., & Murphy, R. (2014). *Learning online: What research tells us about whether, when and how?* Routledge.;

- Mertens, W., & Recker, J. (2020). New guidelines for null hypothesis significance testing in hypothetico-deductive is research. *Journal of the Association for Information Systems*, 21(4), 1.
- Mitchell, C., & Sackney, L. (2001). Communities of leaders: Developing capacity for a learning community. In *Annual Conference of the American Educational Research Association, Seattle, Washington*.
- Mohajan, H. K. (2018). Qualitative research methodology in social sciences and related subjects. *Journal of Economic Development, Environment and People*, 7(1), 23-48.
- Monametsi, J. J. (2015). An analysis of the role of school management teams in teacher professional development (Doctoral dissertation, University of Pretoria).
- Moore-Adams, B. L. (2016, November). Higher Education Faculty Utilization of Online Technological Tools: A Multilevel Analysis. In *E-Learn: World Conference on E-Learning in Corporate, Government, Healthcare, and Higher Education* (pp. 1213-1220). Association for the Advancement of Computing in Education (AACE).
- Moore-Adams, B. L., & Warnick, S. (2016, March). Establishing presence and community in the online classroom. In *Society for Information Technology & Teacher Education International Conference* (pp. 844-849). Association for the Advancement of Computing in Education (AACE).
- Moore-Adams, B. L., Jones, W. M., & Cohen, J. (2016). Learning to teach online: A systematic review of the literature on K-12 teacher preparation for teaching online. *Distance Education*, *37*(3), 333-348.
- Moser, A., & Korstjens, I. (2018). Series: Practical guidance to qualitative research. Part 3: Sampling, data collection and analysis. *European journal of general practice*, 24(1), 9-18.
- Munoz-Chereau, B., & Thomas, S. M. (2016). Educational effectiveness in Chilean secondary education: Comparing different 'value added' approaches to evaluate schools. *Assessment in Education: Principles, Policy & Practice*, 23(1), 26-52.
- Murphy, E., & Rodríguez-Manzanares, M. A. (2009). Teachers' perspectives on motivation in high-school distance education. *International Journal of E-Learning & Distance Education/Revue internationale du e-learning et la formation à distance*, 23(3), 1-24.
- Myende, P. E., & B. S. Nhlumayo, B.S. (2020). "Enhancing Parent-teacher Collaboration in Rural Schools: Parents' Voices and Implications for Schools." *International Journal of Leadership in Education*.

- Naicker, C. (2016). From Marikana to# feesmustfall: The praxis of popular politics in South Africa. *Urbanisation*, 1(1), 53-61.
- Naicker, I., & Somdut, S. (2014). New kids on the block: Novice teachers and teacher leadership. *International Journal of Educational Sciences*, 7(3), 569-578.
- Naidu, S., & Laxman, K. (2019). Factors Inhibiting Teachers' Embracing eLearning in Secondary Education: A Literature Review. *Asian Journal of Distance Education*, *14*(2), 124-143.
- Nappi, J. S. (2014). The teacher leader: Improving schools by building social capital through shared leadership. *Delta Kappa Gamma Bulletin*, 80(4).
- Nartiningrum, N., & Nugroho, A. (2020). Online learning amidst global pandemic: EFL students' challenges, suggestions, and needed materials. *ENGLISH FRANCA:*Academic Journal of English Language and Education, 4(2), 115-140.
- National Council of Teachers of Mathematics (NCTM) (2016),
- Netolicky, D. M. (2020). Transformational professional learning: What, why and how? *Independent Education*, 50(1), 32-33.
- Newton, P., Riveros, A., & da Costa, J. (2013). The Influence of Teacher Leadership in the Career Advancement of Schoolteachers: A Case Study. *EAF Journal*, 23(2). 105.
- Nguyen, T. D., & Hunter, S. (2018). Towards an understanding of dynamics among teachers, teacher leaders, and administrators in a teacher-led school reform. *Journal of Educational Change*, 19(4), 539-565.
- Nieuwenhuis, J. (2007). Introducing qualitative research. In, K. Maree (Ed.), *First steps in Research (4th ed.)*. (pp. 46-68). Pretoria: Van Schaik Publishers
- Norman, T. A. (2020). Methods of a Narrative Inquirist: Storying the Endured Teacher Identity. *The Qualitative Report*, 25(11), 3962-3975.
- Northouse, P. G. (2019). *Introduction to leadership: Concepts and practice*. SAGE Publications.
- Norton, P., & Hathaway, D. (2015). In search of a teacher education curriculum:

 Appropriating a design lens to solve problems of practice. *Educational Technology*, 3-14.
- Nugroho, A., & Atmojo, A. E. P. (2020). Digital learning of English beyond classroom: EFL learners' perception and teaching activities. *JEELS (Journal of English Education and Linguistics Studies)*, 7(2), 219-243.

- Nunes, M., & McPherson, M. (2006). Learning support in online constructivist environments in information systems. *Innovation in Teaching and Learning in Information and Computer Sciences*, 5(2), 1-9
- O'Toole, V. M. (2018). "Running on fumes": emotional exhaustion and burnout of teachers following a natural disaster. *Social Psychology of Education*, *21*(5), 1081-1112.
- Omotayo, F. O., & Haliru, A. (2020). Perception of task-technology fit of digital library among undergraduates in selected universities in Nigeria. *The Journal of Academic Librarianship*, 46(1), 102097.
- Opfer, V. D., Pedder, D. J., & Lavicza, Z. (2011). The influence of school orientation to learning on teachers' professional learning change. *School Effectiveness and School Improvement*, 22(2), 193-214.
- Palmer, C. F. D. (2011). *Insiders' voices: A phenomenological study of informal teacher leadership from the perspective of those who choose to lead* (Doctoral dissertation, University of Pittsburgh).
- Panich, V. (2016). *Entertainment of teachers' life into the learning community*. Bangkok: SR Printing Mass Product
- Pareja Roblin, N. N., Ormel, B. J., McKenney, S. E., Voogt, J. M., & Pieters, J. M. (2014). Linking research and practice through teacher communities: a place where formal and practical knowledge meet? *European Journal of Teacher Education*, *37*(2), 183-203.
- Pareja Roblin, N. N., Ormel, B. J., McKenney, S. E., Voogt, J. M., & Pieters, J. M. (2014). Linking research and practice through teacher communities: a place where formal and practical knowledge meet? *European Journal of Teacher Education*, *37*(2), 183-203.
- Parrish, A. H., & Sadera, W. A. (2019). A review of faculty development models that build teacher educators' technology competencies. *Journal of Technology and Teacher Education*, 27(4), 437-464.
- Patrick, B. C., Hisley, J., & Kempler, T. (2000). "What's everybody so excited about?" The effects of teacher enthusiasm on student intrinsic motivation and vitality. *The Journal of Experimental Education*, 68(3), 217-236.
- Peñarroja, V., Orengo, V., Zornoza, A., & Hernández, A. (2013). The effects of virtuality level on task-related collaborative behaviors: The mediating role of team trust. *Computers in Human Behavior*, 29(3), 967-974.
- Penuel, W. R., Fishman, B. J., Yamaguchi, R., & Gallagher, L. P. (2007). What makes professional development effective? Strategies that foster curriculum implementation. *American Educational Research Journal*, 44, 951-958

- Penuel, W. R., Fishman, B. J., Yamaguchi, R., & Gallagher, L. P. (2007). What makes professional development effective? Strategies that foster curriculum implementation. *American Educational Research Journal*, 44(4), 921-958.
- Petty, N. J., Thomson, O. P., & Stew, G. (2012). Ready for a paradigm shift? Part 2: Introducing qualitative research methodologies and methods. *Manual therapy*, 17(5), 378-384.
- Pham, L. T. M. (2018). Qualitative approach to research a review of advantages and disadvantages of three paradigms: Positivism, interpretivism and critical inquiry. *University of Adelaide*.
- Powell, C. G., & Bodur, Y. (2019). Teachers' perceptions of an online professional development experience: Implications for a design and implementation framework. *Teaching and Teacher Education*, 77, 19-30.
- Prensky, M. (2001). Digital natives, digital immigrants. On the Horizon, 9(5), 1–2
- Printy, S., Otten, L., Krantz-Selleck, K., & Usiak, M. (2008). *Leadership content knowledge:* what do we know? Working Papers for K-12 Educational Administration.
- Punch, K. F., & Oancea, A. (2014). *Introduction to research methods in education*. SAGE Publishers.
- Rajagopaul, S. M. (2007). An investigation into the factors that help or hinder teacher leadership: case studies of three urban primary schools in the Pietermaritzburg region (Doctoral dissertation).
- Rasmitadila, R., Widyasari, W., Humaira, M., Tambunan, A., Rachmadtullah, R., & Samsudin, A. (2020). Using blended learning approach (BLA) in inclusive education course: A study investigating teacher students' perception. *International Journal of Emerging Technologies in Learning (IJET)*, 15(2), 72-85
- Ratts, R. F., Pate, J. L., Archibald, J. G., Andrews, S. P., Ballard, C. C., & Lowney, K. S. (2015). The influence of professional learning communities on student achievement in elementary schools. *Journal of Education & Social Policy*, 2(4), 51-61.
- Reimers, F., & Schleicher, A. (2020). Schooling disrupted, schooling rethought. *How the COVID-19 Pandeminc is Changing Education. Retrieved December*, *14*, 2020.
- Republic of South Africa 1 996a. *South African Schools Act* (Act 84 of 1996). Pretoria: Government Printers.
- Ribeiro, M. D. (Ed.). (2020). Examining social identities and diversity issues in group therapy: Knocking at the boundaries. Routledge.

- Rottbauer, W., Kessler, M., Mathew Williams, P., Mahoney, R. S. V. B., Price, M. J., Grasso, C., & Kar, S. (2020). Contemporary clinical outcomes with MitraClipTM (NTR/XTR) system: core-lab echo results from+ 1000 patient the Global EXPAND study. PCR, editor. Pcronline. com 2020.W. Kidd and J. Murray, "The Covid-19 pandemic and its effects on teacher education in England:
- Rule, P., & John, V. (2011). Your guide to case study research. Pretoria: van Schaik.
- Ruthven, K., Hennessy, S., & Brindley, S. (2004). Teacher representations of the successful use of computer-based tools and resources in secondary-school English, Mathematics and Science. *Teaching and Teacher Education*, 20(3), 259-275.
- Saldana, J. (2009). *The coding manual for qualitative researchers*. Thousand Oaks, CA: Sage Publications, Ltd.
- Sargent, T. C., & Hannum, E. (2009). Doing more with less: Teacher professional learning communities in resource-constrained primary schools in rural China. *Journal of Teacher Education*, 60(3), 258-276.
- Sarkar (2012 Sarkar, S. (2012). The role of information and communication technology (ICT) in higher education for the 21st century. *Science*, *1*(1), 30-41.
- Schmidt, S. W., Tschida, Christina, M., & Hodge, E. M. (2016). How Faculty Learn to Teach Online: What Administrators Need to Know. *Distance Education Report*, 20(9), 1–3.
- Schott, C., van Roekel, H., & Tummers, L. G. (2020). Teacher leadership: A systematic review, methodological quality assessment and conceptual framework. *Educational Research Review*, *31*, 100352.
- Shaked, H., Schechter, C., Ganon-Shilon, S., & Goldratt, M. (2017). *Systems Thinking for School Leaders*. Dordrecht: Springer.
- Shariman, T. P. N. T., Razak, N. A., & Noor, N. F. M. (2012). Digital literacy competence for academic needs: An analysis of Malaysian students in three universities. *Procedia-Social and Behavioral Sciences*, 69, 1489-1496.
- Shava, G. N., & Tlou, F. N. (2018). Distributed Leadership in Education, Contemporary Issues in Educational Leadership. *African Educational Research Journal*, 6(4), 279-287.
- Shen, C. W., & Ho, J. T. (2020). Technology-enhanced learning in higher education: A bibliometric analysis with latent semantic approach. *Computers in Human Behavior*, 104, 106177.
- Shepherd-Jones, A. R., & Salisbury-Glennon, J. D. (2018). Perceptions matter: the correlation between teacher motivation and principal leadership styles. *Journal of Research in Education*, 28(2), 93-131.

- Shindler, J., & Fleisch, B. (2007). Closing the gap on schooling for all in South Africa. *International Review of Education*, 53, 135-157.
- Singh, A., Yager, S. O., Yutakom, N., Yager, R. E., & Ali, M. M. (2012). Constructivist Teaching Practices Used by Five Teacher Leaders for the Iowa Chautauqua Professional Development Program. *International Journal of Environmental and Science Education*, 7(2), 197-216.
- Smulyan, L. (2016). Symposium introduction: Stepping into their power: The development of a teacher leadership stance. *Schools*, *13*(1), 8-28.
- Spillane, J. P., & Coldren, A. F. (2015). *Diagnosis and design for school improvement: Using a distributed perspective to lead and manage change*. Teachers College Press.
- Spillane, J. P., Halverson, R., & Diamond, J. B. (2001). Investigating school leadership practice: A distributed perspective. *Educational researcher*, *30*(3), 23-28.
- Stein, K. C., Macaluso, M., & Stanulis, R. N. (2016). The interplay between principal leadership and teacher leader efficacy. *Journal of School Leadership*, 26(6), 1002-1032.
- Strunc, A. R. (2020). Are They Listening? Policymakers and Their Role in Public Education. *Research in Educational Policy and Management*, 2(1), i-iii.
- Thapa, A., Cohen, J., Guffey, S., & Higgins-D'Alessandro, A. (2013). A review of school climate research. Review of educational research, 83(3), 357-385.
- The World Bank. (2020). *Rapid Response Briefing Note: Remote Learning and COVID-19 Outbreak*. The World Bank: Washington, DC, USA, 2020; pp. 1–10.
- Thompson, M., Ponte, E., Paek, P., & Goe, L. (2004). Assessment and Support System for Teachers: Methodological considerations and recommendations for future research. Study of the Impact of the California Formative, 1-47.
- Thornton, K., & Cherrington, S. (2014). Leadership in professional learning communities. *Australasian Journal of Early Childhood*, *39*(3), 94-102.
- Tomkins, L. (2020). Where is Boris Johnson? When and why it matters that leaders show up in a crisis. *Leadership*, *16*(3), 331-342
- Trust, T., & Whalen, J. (2021). Emergency remote teaching with technology during the COVID-19 pandemic: using the whole teacher lens to examine educator's experiences and insights. *Educational Media International*, 58(2), 145-160.
- Van Wart, M., Roman, A., Wang, X., & Liu, C. (2017). Integrating ICT adoption issues into (e-) leadership theory. *Telematics and Informatics*, *34*(5), 527-537.

- Von Dohlen, H. B., & Karvonen, M. (2018). Teachers' Self-Reported Leadership Behaviors in Formal and Informal Situations. *International Journal of Teacher Leadership*, 9(2), 69-89.
- Vongkulluksn, V. W., Xie, K., & Bowman, M. A. (2018). The role of value on teachers' internalization of external barriers and externalization of personal beliefs for classroom technology integration. *Computers & Education*, 118, 70-81.
- Wanzare, Z. (2012). Instructional supervision in public secondary schools in Kenya. *Educational Management Administration & Leadership*, 40(2), 188-216.
- Watson, E. (2020). STEM or STEAM? The critical role of arts in technology education (and the critical role of art in technology). *Irish Journal of Academic Practice*, 8(1), 8.
- Watson, J. S. (2014). Assessing creative process and product in higher education. *Practitioner Research in Higher Education*, 8(1), 89-100.
- Webber, C. F. (2021). The need for cross-cultural exploration of teacher leadership. *Research* in Educational Administration and Leadership, 6(1), 17-49
- Webber, C. F., & Nickel, J. (2021). Sustainable Teacher Leadership. *Research in Educational Administration & Leadership*, 6(1), 277-311.
- Wenner, J. A., & Campbell, T. (2017). The theoretical and empirical basis of teacher leadership: A review of the literature. *Review of Educational Research*, 87(1), 134-171.
- Willis, J. (1995). A recursive, reflective instructional design model based on constructivist-interpretivist theory. *Educational Technology*, *35*(6), 5-23.
- Wilson, A. (2016). From professional practice to practical leader: Teacher leadership in professional learning communities. *International Journal of Teacher Leadership*, 7(2), 45-62.
- Yendol-Hoppey (2010Yendol-Hoppey, D., & Dana, N. F. (Eds.). (2010). Powerful professional development: Building expertise within the four walls of your school. Corwin Press.
- Yin, H., To, K. H., Keung, C. P. C., & Tam, W. W. Y. (2019). Professional learning communities count: Examining the relationship between faculty trust and teacher professional learning in Hong Kong kindergartens. *Teaching and Teacher Education*, 82, 153–163.
- York-Barr & Duke, 2004). York-Barr, J., & Duke, K. (2004). What do we know about teacher leadership? Findings from two decades of scholarship. *Review of Educational Research*, 74(3), 255-316.

- Yuen, A. H., & Ma, W. W. (2008). Exploring teacher acceptance of e-learning technology. *Asia-Pacific Journal of Teacher Education*, 36(3), 229-243.
- Zahroh, A. (2015). Membangun Kualitas Pembelajaran Melalui Dimensi Profesionalisme [Professionalism]. Bandung: Yrama Widya.
- Zhong, L. (2017). Indicators of digital leadership in the context of K-12 education. *Journal of Educational Technology Development and Exchange (JETDE)*, 10(1), 3.

APPENDIX A: PERMISSION LETTER TO KZN DoE

12John Rose Drive

Mariann Heights

3610

26 February 2021

Dear Sir/Madam

anyone.

REQUEST FOR PERMISSION TO CONDUCT RESEARCH AT YOUR SCHOOL

My name is Ngoako Gloria Khululiwe and I am conducting a research as a requirement of the University of KwaZulu-Natal towards a Degree of Masters in Education. The title of the research study is "The role of teacher leaders in developing and leveraging professional

learning communities during and post the COVID-19 pandemic".

I would like to use your school as one of the research sites, and this letter intends to request your permission. The focus of the study is on teacher leaders therefore I would like to request your teachers to participate in the study. Should permission be granted, the interviews with the ... will be scheduled for dates and times that are convenient for them. Care will be taken that no disruption is caused during such interviews. Please also note that the participation in this study is voluntary, and the participant has the right to withdraw from the study at any time without any negative consequence. In addition, you are assured that details of the school and the participant will be kept confidential, and your identity will never be disclosed to

For more information and questions about the study, you may contact the researcher or the research supervisor on the following details:

Name of researcher: Ngoako Gloria Khululiwe.

Email: gkhululiwengoako@gmaill.com

Supervisor: Dr. BNCK Mkhize: Tel No.: (031) 260 1870; Email: Mkhizeb3@ukzn.ac.za

You may also contact the Research Office through:

Govan Mbeki Building
Westville Campus
HSSREC@ukzn.ac.za
Research Office,
Tel.: 031 260 4557 E-mail: mohunp@ukzn.ac.za
Thanking you in advance.

Yours in Education

Mrs. G.K Ngoako

APPENDIX B: PERMISSION LETTER TO SCHOOL PRINCIPALS

12 John Rose Drive

Mariann Heights

3610

26 February 2021

Dear Sir/Madam

REQUEST FOR PERMISSION TO CONDUCT RESEARCH AT YOUR SCHOOL

My name is Ngoako Gloria Khululiwe and I am conducting a research as a requirement of the

University of KwaZulu-Natal towards a Degree of Masters in Education. The title of the

research study is "The role of teacher leaders in developing and leveraging professional

learning communities during and post the COVID-19 pandemic".

I would like to use your school as one of the research sites, and this letter intends to request

your permission. The focus of the study is on teacher leaders therefore I would like to request

your teachers to participate in the study. Should permission be granted, the interviews with

the ... will be scheduled for dates and times that are convenient for them. Care will be taken

that no disruption is caused during such interviews. Please also note that the participation in

this study is voluntary, and the participant has the right to withdraw from the study at any

time without any negative consequence. In addition, you are assured that details of the school

and the participant will be kept confidential, and your identity will never be disclosed to

anyone.

Your positive response in this regard will be highly appreciated. Should you need further

information regarding the matter, you may contact me, my supervisor and the Research

Office of the University.

CONTACT DETAILS

Researcher: Ngoako Gloria Khululiwe

Email: gkhululiwengoako@gmail.com

Supervisor: Dr BNCK Mkhize,

Email: Mkhizeb3@ukzn.ac.za

94

Telephone: 031 260 1398

You may also contact the Research Office through:

Govan Mbeki Building

Westville Campus

HSSREC Research Office,

Tel.: 031 260 8350/4557/3587

E-mail: hssrec@ukzn.ac.za

Yours sincerely,

Ngoako Gloria Khululiwe

APPENDIX C: PERMISSION LETTER TO PARTICIPANTS

12 John Rose Drive

Mariann Heights

Your participant

REQUEST FOR PARTICIPATION IN THE RESEARCH

My name is Gloria Khululiwe Ngoako and I am conducting a research as a requirement at the University of KwaZulu-Natal towards a Degree of Masters in Education. The title of the research is "The role of teacher leaders in developing and leveraging professional learning communities during and post the COVID-19 pandemic". The objectives of the study are:

- To investigate the role of teacher leaders in developing and leveraging professional learning communities in their schools during and post the COVID-19 pandemic
- To explore the barriers that teacher leaders may encounter when developing and leveraging PLCs in their schools during and post the COVID-19 pandemic
- To explore strategies that teacher leaders may use to mitigate the barriers that they may encounter when developing and leveraging PLC in their schools during and post COVI-19 pandemic

The study will focus on teacher leaders. This letter intends to elucidate the purpose of the study and to request your participation in the study.

Please note that:

- Your confidentiality is guaranteed as your inputs will not be attributed to you in person, but reported only as a population member opinion.
- The interview may last for about 1 hour and may be split into two parts depending on your preference.
- Any information given by you cannot be used against you, and the collected data will be used for purposes of this research only.
- Data will be stored in secure storage and destroyed after 5 years.
- You have a choice to participate, not participate or stop participating in the research. You will not be penalised zed for taking such an action.
- Your involvement is purely for academic purposes only, and there are no financial benefits involved.

• If you are willing to be interviewed, please indicate (by ticking as applicable) whether or not you are willing to allow the interview to be recorded by the following equipment:

	Willing	Not willing
Audio equipment		

I can be contacted at:

Email: gkhululiwengoako@gmail.com

Cell: 0723031603

My supervisor is Dr. BNCK Mkhize who is located at the School of Education, at the University of KwaZulu-Natal. He can be contacted at:

Email: Mkhizeb3@ukzn.ac.za

Phone: 031 260 1870

I hope this letter will find your positive consideration, thanking you in advance.

Yours Sincerely

Ngoako Gloria Khululiwe

APPENDIX D: DECLARATION

Declaration

PLEASE COMPLETE THE FOLLOW PARTICIPATION:	WING SECTION FOR CONSENT OF
I	(Full names of participant) hereby confirm
that I understand the nature and purpo	se of the study entitled: "The role of teacher leaders in
developing and leveraging professiona	al learning communities during and post the COVID-19
pandemic". I agree to participate in the	ne study. I am also fully aware that I have the right to
withdraw from the study at any point	should I wish to do so, without any negative or
undesirable consequence. I am also av	ware that there are neither any foreseeable direct
benefits nor direct risks associated wit	th my participation in this study. I therefore understand
the contents of this letter fully and I do	o GIVE CONSENT / DO NOT GIVE CONSENT for
the interviews to be digitally recorded	
Signature	Date

APPENDIX E: INTERVIEW SCHEDULE RESEARCH INSTRUMENTS

Interview Schedule

When conducting the interviews, the open-ended questions will be asked. The interview will be in one day but it will be spread in two sessions with two participants per session.

Semi-structured interview

Interviews will be conducted through zoom.

- 1. What do you understand about professional learning communities (PLCs)?
- 2. As a teacher leader what is your role in developing PLC in your school?
- 3. How has COVID-19 impacted on the functioning of PLCs in your school? Please elaborate. Probes: Please comment on how teachers interact and collaborate with each other in PLCs during this time. Are you able to foster an atmosphere of trust? Do you continue to monitor the work of PLCs and provide constructive feedback? Are you able to address the most pressing instructional challenges in your PLCs?
- 4. As a teacher leader, are you able to promote coherent and connected opportunities for teacher learning in your PLCs during this pandemic? Please elaborate. Probes: If so, how? If no, Why?
- 5. What challenges (if any) have you encountered in developing PLC during this time? Please elaborate.
- 6. What means or strategies have you tried to leverage PLCs during the pandemic? Probes: have you tried using digital networks? If so, what are the benefits and challenges of using digital networks to leverage PLCs?
- 7. With the COVID-19 protocols in place, how have you tried to 'deprivatise practice' and prevent teacher isolation?
- 8. To what extent are you able to maintain shared routines in your PLCs?
- 9. What do you think is the way forward for PLCs post the pandemic?
- 10. What other information can you add that I may not have asked you about your role in developing and leveraging PLCs during and beyond the COVID-19 pandemic?

APPENDIX F: TURNITIN CERTIFICATE



APPENDIX G: ETHICAL CLEARANCE CERTIFICATE



14 December 2021

Gloria Khululiwe Ngoako (203516362) School Of Education **Edgewood Campus**

Dear GK Ngoako,

Protocol reference number: HSSREC/00003651/2021

Project title: The role of teacher leaders in developing and leveraging professional learning communities during

and post the COVID-19 pandemic.

Degree: Masters

Approval Notification - Expedited Application

This letter serves to notify you that your application received on 24 November 2021 in connection with the above, was reviewed by the Humanities and Social Sciences Research Ethics Committee (HSSREC) and the protocol has been granted FULL APPROVAL.

Any alteration/s to the approved research protocol i.e. Questionnaire/Interview Schedule, Informed Consent Form, Title of the Project, Location of the Study, Research Approach and Methods must be reviewed and approved through the amendment/modification prior to its implementation. In case you have further queries, please quote the above reference number. PLEASE NOTE: Research data should be securely stored in the discipline/department for a period of 5 years.

This approval is valid until 14 December 2022.

To ensure uninterrupted approval of this study beyond the approval expiry date, a progress report must be submitted to the Research Office on the appropriate form 2 - 3 months before the expiry date. A close-out report to be submitted when study is finished.

All research conducted during the COVID-19 period must adhere to the national and UK2N guidelines.

HSSREC is registered with the South African National Research Ethics Council (REC-040414-040).

Yours sincerely,



Professor Dipane Hlalele (Chair)

/dd

Humanities and Social Sciences Research Ethics Committee

Postal Address: Private Bag X54001, Durban, 4000, South Africa

Telephone: +27 (0)31 260 6150/4557/1567 Ersall: harvojjukzn.ac.za Website: http://www.chukzn.ac.za/Research-Ethica

Founding Compuses: - Bágewood

Howard College

- Medical School - Fleterworksburg

INSPIRING GREATNESS